9 18 052

# OTIC FILE COP.

#### A COMPARATIVE ANALYSIS OF THE MEDICAL SUPPORT IN THE COMBAT OPERATIONS IN THF FALKLANDS CAMPAIGN AND THE GRENADA EXPEDITION

A thesis presented to the Faculty of the U.S. Army Command and General Staff College in partial fulfillment of the requirements for the degree

×. ...

MASTER OF MILITARY ART AND SCIENCE

by

THOMAS E. BROYLES, MAJ, USA B.A., Stetson University, 1973 M.H.A., Baylor University, 1982

> Fort Leavenworth, Kansas 1987

Approved for public release distribution is unlimited.

87



87-3585

		•	REPORT DOCU	MENTATION	PAGE		
1a. REPORT	SECURITY CLASS	SIFICATION		1b. RESTRICTIVE			
	lassified						
	Y CLASSIFICATIO		:	3. DISTRIBUTION	N/AVAILABILITY		•
2b. DECLAS	SIFICATION / DOW	VNGRADING SCHED	ULE	, onerassi	rited/ourth	it cea	
		ION REPORT NUME	••	5. MONITORING	ORGANIZATION	REPORT NUME	BER(S)
	f College	and and Gener	ra]				
	F PERFORMING	ORGANIZATION	6b. OFFICE SYMBOL (If applicable)	7a. NAME OF N	IONITORING OR	GANIZATION	
		n, KS 66027-					1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 -
6c. ADDRESS	i (City, State, and	d ZIP Code) 6900		7b. ADDRESS (C	ity, State, and Z	IP Code)	
×		•	•		:	. •	•
0- 11440 0							
ORGANIZ	F FUNDING/SPOI	NSORING	8b. OFFICE SYMBOL (if applicable)	9. PROCUREMEN	IT INSTRUMENT	IDENTIFICATION	INUMBER
	1011						يور حيد المن وحياة ما المركب المركب
OC: ADDRESS	(City, State, and	ZIP Code)		10. SOURCE OF	PROJECT	TASK	WORK UNIT
•			•	ELEMENT NO.	NO.	NO.	ACCESSION N
11. HILE (IA Δ ຕິດ)	CIUCHE SECURITY CI	ASSIFICATION)			_		
FALK	LANDS CAMPA	IGN AND THE	HE MEDICAL SUPPO GRENADA EXPEDITI	ON	MBAI OPERA	TIONS IN TH	iE ·
	R THOMAS E.	BROVI EC	· · · · · · · · · · · · · · · · · · ·		······································		
13a. TYPE O		13b. TIME C	OVERED	14 DATE OF DES			
	er's Thesis		-1986 TO 6-1987	14. DATE OF REPO		<b>n, Day)</b> 15. PA 15	GE COUNT
16 CLIDDI ENA	ENTARY NOTATI	ON				والمرجع والمرجع والمحار المحار المحار المحار المحار المحار المحار والمحار والمحار والمحار والمحار والمحار والم	
IV. JUPPEEM							
		· · · · · · · · · · · · · · · · · · ·					
17.	COSATI C	and the second	18. SUBJECT TERMS (				
17. FIELD	COSATI C GROUP	ODES SUB-GROUP	AN ANALYSIS	OF MEDICAL S	SUPPORT OF	RAPIDLY DE	
17. FIELD	GROUP	SUB-GROUP	AN ANALYSIS FORCES (in t	OF MEDICAL Sthe Falklands	SUPPORT OF	RAPIDLY DE	
17. FIELD 19. ABSTRAC	GROUP T (Continue on n	SUB-GROUP	AN ANALYSIS FORCES (in f	OF MEDICAL s the Falklands	SUPPORT OF 5 and in Gr	RAPIDLY DE enada)	PLOYED LAND
17. FIELD 19. ABSTRAC This Falklan	GROUP T (Continue on n study exami ds Campaign	SUB-GROUP everse if necessary ines the medi n of 1982 and	AN ANALYSIS FORCES (in f and identify by block of cal support of c	OF MEDICAL s the Falklands w <b>mber</b> ) combat operat	SUPPORT OF s and in Gr cions by th	RAPIDLY DE enada) e British	PLOYED LAND
17. FIELD 19. ABSTRAC This FalkJan Kecent	GROUP T (Continue on n study exami ds Campaign history por	SUB-GROUP everse if necessary ines the medi n of 1982 and stends the in	AN ANALYSIS FORCES (in f and identify by block m cal support of c by the American creased use of a	OF MEDICAL s the Falklands combat operat is in the Gre	SUPPORT OF s and in Gr cions by th enada Exped	RAPIDLY DE enada) e British ition of 1	PLOYED LAND in the 983.
17. FIELD 19. ABSTRAC This Falklan Kecent employm	GROUP T (Continue on on study exami ds Campaign history por ents far fr	SUB-GROUP everse if necessary ines the medi n of 1982 and rtends the in rom logistica	AN ANALYSIS FORCES (in f and identify by block of cal support of c by the American acceased use of g l bases. Medica	OF MEDICAL s the Falklands combat operat is in the Gre ground combat	SUPPORT OF s and in Gr zions by th enada Exped ; forces in	RAPIDLY DE enada) e British ition of 1 short dura	PLOYED LAND in the 983. ation
17. FIELD 19. ABSTRAC This FalkJan Kecent employm and by tions w	GROUP T (Continue on or study exami ds Campaign history por ents far fr the America ith each ot	SUB-GROUP everse if necessary ines the medi- n of 1982 and rtends the in rom logistica ans in Granad ther and agai	AN ANALYSIS FORCES (in f and identify by block of cal support of of by the American acceased use of g l bases. Medica a is analyzed by past established	OF MEDICAL S the Falklands combat operat is in the Gre ground combat i support by comparing t	SUPPORT OF s and in Gr tions by th enada Exped t forces in the Briti. the two med	RAPIDLY DE enada) e British ition of 1 short dura sh in the 1 ical suppor	PLOYED LAND in the 983. ation Falklands rt opera-
17. FIELD 19. ABSTRAC This Falklan Kecent employm and by tions w medical	GROUP GROUP T (Continue on a study exami ds Campaign history por ents far fr the America ith each ot support pr	SUB-GROUP everse if necessary ines the medi- n of 1982 and rtends the in rom logistica ans in Grenad ther and agai- rinciples and	AN ANALYSIS FORCES (in f and identify by block of cal support of c by the American increased use of g l bases. Medica a is analyzed by nst established operational con	OF MEDICAL s the Falklands combat operat is in the Gre ground combat of support by comparing t principles a cents identi	SUPPORT OF s and in Gr s and i	RAPIDLY DE enada) e British ition of 1 short dura sh in the i ical support onal concept	PLOYED LAND in the 983. ation Falklands rt opera- ots. The
17. FIELD 19. ABSTRAC This FalkJan Recent employm and by tions w medical tools f	GROUP GROUP T (Continue on a study exami ds Campaign history por ents far fr the America ith each ot support pr or comparin	SUB-GROUP everse if necessary ines the medi- n of 1982 and rtends the in- rom logistica ans in Granad ther and agai- rinciples and ng the two me	AN ANALYSIS FORCES (in f and identify by block of cal support of c by the American acceased use of g bases. Medica a ls analyzed by nst established operational con dical support on	OF MEDICAL s the Falklands combat operat is in the Gre ground combat il support by comparing t principles a cepts identi erations stu	SUPPORT OF s and in Gr cions by th enada Exped torces in the Briti the two med ind operation fied are so	RAPIDLY DE enada) e British : ition of 1 short dura sh in the i ical suppor onal concep hown to be	PLOYED LAND in the 983. ation Falklands rt opera- ots. The excellent
17. FIELD 19. ABSTRAC This Falklan Recent employm and by tions w medical tools f	GROUP GROUP T (Continue on a study exami ds Campaign history por ents far fr the America ith each ot support pr or comparin	SUB-GROUP everse if necessary ines the medi- n of 1982 and rtends the in- rom logistica ans in Granad ther and agai- rinciples and ng the two me	AN ANALYSIS FORCES (in f and identify by block of cal support of c by the American increased use of g l bases. Medica a is analyzed by nst established operational con	OF MEDICAL s the Falklands combat operat is in the Gre ground combat il support by comparing t principles a cepts identi erations stu	SUPPORT OF s and in Gr cions by th enada Exped torces in the Briti the two med ind operation fied are so	RAPIDLY DE enada) e British : ition of 1 short dura sh in the i ical suppor onal concep hown to be	PLOYED LAND in the 983. ation Falklands rt opera- ots. The excellent
17. FIELD 19. ABSTRAC This FalkJan Recent employm and by tions w medical tools f	GROUP GROUP T (Continue on a study exami ds Campaign history por ents far fr the America ith each ot support pr or comparin	SUB-GROUP everse if necessary ines the medi- n of 1982 and rtends the in- rom logistica ans in Granad ther and agai- rinciples and ng the two me	AN ANALYSIS FORCES (in f and identify by block of cal support of c by the American acceased use of g bases. Medica a ls analyzed by nst established operational con dical support on	OF MEDICAL s the Falklands combat operat is in the Gre ground combat il support by comparing t principles a cepts identi erations stu	SUPPORT OF s and in Gr cions by th enada Exped torces in the Briti the two med ind operation fied are so	RAPIDLY DE enada) e British : ition of 1 short dura sh in the i ical suppor onal concep hown to be	PLOYED LAND in the 983. ation Falklands rt opera- ots. The excellent
17. FIELD 19. ABSTRAC This Falklan Recent employm and by tions w medical tools f	GROUP GROUP T (Continue on a study exami ds Campaign history por ents far fr the America ith each ot support pr or comparin	SUB-GROUP everse if necessary ines the medi- n of 1982 and rtends the in- rom logistica ans in Granad ther and agai- rinciples and ng the two me	AN ANALYSIS FORCES (in f and identify by block of cal support of c by the American acceased use of g bases. Medica a ls analyzed by nst established operational con dical support on	OF MEDICAL s the Falklands combat operat is in the Gre ground combat il support by comparing t principles a cepts identi erations stu	SUPPORT OF s and in Gr cions by th enada Exped torces in the Briti the two med ind operation fied are so	RAPIDLY DE enada) e British : ition of 1 short dura sh in the i ical suppor onal concep hown to be	PLOYED LAND in the 983. ation Falklands rt opera- ots. The excellent
17. FIELD 19. ABSTRAC This Falklan Kecent employm and by tions w medical tools f and exe	GROUP T (Continue on or study exami ds Campaign history por ents far fr the America ith each ot support pr or comparin cuting futu	SUB-GROUP everse if necessary ines the medi- ne of 1982 and thends the in- com logistica ans in Granad ther and agai- ther and agai- ther and agai- ther two me- are medical s	AN ANALYSIS FORCES (in f and identify by block of cal support of c by the American acceased use of g bases. Medica a ls analyzed by nst established operational con dical support on	OF MEDICAL s the Falklands combat operat s in the Gre ground combat of support by comparing t principles a cepts identi erations stu s of rapidly	SUPPORT OF s and in Gr s and in Gr s and in Gr s and Exped s forces in the Briti the Briti the two med and operation fied are sh died as we deployed	RAPIDLY DE enada) e British ition of 1 short dura sh in the i ical support onal concept hown to be 11 as for p land forces	PLOYED LAND in the 983. ation Falklands rt opera- ots. The excellent
17. FIELD 19. ABSTRAC This Fa]k]en Kecent employm and by tions w medical too]s f and exe 20. DISTRIBU UNCLAS	GROUP T (Continue on on study exami- ids Campaign history por ents far fr the America ith each ot support pr or comparin cuting futu	SUB-GROUP everse if necessary ines the medi- n of 1982 and rtends the in- rom logistica ans in Granad ther and agai- rinciples and ag the two me are medical s	AN ANALYSIS FORCES (in f and identify by block of cal support of c by the American acceased use of g l bases. Medica a is analyzed by nst established operational con dical support op upport operation	OF MEDICAL S the Falklands combat operat is in the Gre ground combat il support by comparing t principles a cepts identi erations stu s of rapidly 21. ABSTRACT SEC Unclass	SUPPORT OF s and in Gr s and in Gr cions by th enada Exped torces in the Briti the two med nd operation fied are so died as we died as we deployed	RAPIDLY DE enada) e British ition of 1 short dura sh in the 1 ical support onal concep hown to be 11 as for p land forces CATION	PLOYED LAND in the 983. ation Falklands rt opera- ots. The excellent olanning
17. FIELD 19. ABSTRAC This Fa]k]an Kecent employm and by tions w medical too]s f and exe 20. DISTRIBU DUNCLAS	GROUP T (Continue on or study exami ds Campaign history por ents far fr the America ith each ot support pr or comparin cuting futu	SUB-GROUP everse if necessary ines the medi- n of 1982 and rtends the in- rom logistica ans in Granad ther and agai- rinciples and ag the two me are medical s	AN ANALYSIS FORCES (in f and identify by block of cal support of c by the American acceased use of g l bases. Medica a is analyzed by nst established operational con dical support op upport operation	OF MEDICAL S the Falklands combat operat s in the Gre ground combat of support by comparing t principles a cepts identi erations stu s of rapidly 21. ABSTRACT SEC	SUPPORT OF s and in Gr s and in Gr cions by th enada Exped torces in the Briti the two med nd operation fied are so died as we died as we deployed	RAPIDLY DE enada) e British ition of 1 short dura sh in the 1 ical support onal concep hown to be 11 as for p land forces CATION	PLOYED LAND in the 983. ation Falklands rt opera- ots. The excellent olanning
17. FIELD 19. ABSTRAC This Falklan Kecent employm and by tions w medical tools f and exe 20. DISTRIBU UNCLAS 223. NAME C	GROUP T (Continue on on study exami- ids Campaign history por ents far fr the America ith each ot support pr or comparin cuting futu	SUB-GROUP everse if necessary ines the medi- n of 1982 and rtends the in- rom logistica ans in Granad ther and agai- rinciples and hg the two me are medical s	AN ANALYSIS FORCES (in f and identify by block of cal support of c by the American acceased use of g l bases. Medica a is analyzed by nst established operational con dical support op upport operation	OF MEDICAL S the Falklands combat operat is in the Gre ground combat il support by comparing t principles a cepts identi erations stu s of rapidly 21. ABSTRACT SEC Unclass 22b. TELEPHONE (	SUPPORT OF s and in Gr s and in Gr s and in Gr s and in Gr s the Briti the two med and operation fied are so died as we deployed for CURITY CLASSIFI sified	RAPIDLY DE enada) e British ition of 1 short dura sh in the i ical support onal concept hown to be ll as for p land forces CATION	PLOYED LAND in the 983. ation Falklands rt opera- ots. The excellent olanning

#### MASTER OF MILITARY ART AND SCIENCE

#### THESIS APPROVAL PAGE

Name of Candidate: MAJ Thomas B. Broyles

Title of Thesis: A Comparative Analysis of the Medical Support in the Combat Operations in the Falklands Campaign and the Grenada Expedition

Approved By:

Thesis Committee Chairman LTC Scottie T. Hooker, MBA

Die Deotere I. Hooker, MDA

, Member, Graduate Faculty LTC Johnson, MHA \_\_\_\_, Member, Consulting Faculty -----Rh.D. COL Max G. Manwaring, Accepted this 5th day of June 1987 by: \_, Director, Graduate Degree Brookes, Philip J. Ph.D. Programs

The opinions and conclusions expressed herein are those of the student author and do not necessarily represent the views of the U.S. Army Command and General Staff College or any other governmental agency. (Reference to this study should include the foregoing statement.)

NTIS CRA&I DTIC TAB Undersourced Justef Lettern	<b>N</b> 	QUALITY
By Dectado trony		INSPECTED
198.47 (21)		
D.A. State		}
A-1		ن ن

A START START STARTER CONTRACTOR

#### ABSTRACT

A COMPARATIVE ANALYSIS OF THE MEDICAL SUPPORT IN THE COMBAT OPERATIONS IN THE FALKLANDS CAMPAIGN AND THE GRENADA EXPEDITION, by Major Thomas E. Broyles, USA, 156 pages.

This study examines the medical support of combat operations by the British in the Falklands Campaign of 1982 and by the Americans in the Grenada Expedition of 1983. Recent history portends the increased use of ground combat forces in short duration employments far from logistical bases. These two examples of rapidly deployed land forces are investigated to illustrate the principles and operational concepts of medical support shared in common by both the American and British Armies and needed to support rapidly deployed forces. Medical support by the British in the Falklands and by the Americans in Grenada is analyzed by comparing the two medical support operations with each other and against established principles and operational concepts.

The medical support principles used for this comparative analysis are conformity, provimity, flexibility, mobility, continuity, and control. The medical support operational concepts used include triage, echelons of medical support, elements of combat medicine, patient evacuation, and command and control.

The medical support operations in the Falklands, like those in Grenada, are examined both in planning and execution as well as with respect to the role of medical command and control. Lessons learned by the respective medical departments are discussed in the light of the specific combat operation and in their particular relevance to general rapid deployment force medical support requirements.

The Falklands Campaign points out the highly successful joint medical support provided by the British and the precision with which they planned their medical support operation. It also points out British problems with aeromedical evacuation. The Grenada Expedition illustrates the hazards of precluding medical participation in planning a combat operation and the severe impact on field medical support when a joint operation is poorly coordinated. Both operations reveal the vital role that medical commanders have to play in the timely provision of medical support.

The medical support principles and operational concepts identified are shown to be excellent tools for comparing the two medical support operations studied as well as for planning and executing future medical support operations of rapidly deployed land forces. The lessons derived from each operation, especially when compared with the lessons of the other, also provide vital answers to questions about how medical commanders can ensure their units are ready to medically support combat operations of rapidly deployed forces.

#### **ACKNOWLEDGEMENTS**

From its inception this thesis enjoyed the support of a number of people without whose assistance its completion would not have been possible. First and foremost have been the members of the thesis committee. The chairman, LTC Scottie Hooker, has consistently provided encouragement and motivation plus a discerning critical eye which has been invaluable. LTC Nick Johnson has given me many important insights into the Army Medical Department's operational concepts, especially in the area of aeromedevac. COL Max Manwaring contributed most significantly in assisting me in defining my topic in the formative stages of this project and that has made the thesis writing achievable and enjoyable in the limited time available.

My research about the Falklands would have been impossible without the help of COL R.J.B. Heard, British Liaison Officer at the USACGSC, and COL Ian Creamer, British Liaison Officer, Office of the Surgeon General, U.S. Army. Their assistance in helping me to understand British combat service support and medical support and providing material on those subjects was vital. Their patience and graciousness are deeply appreciated.

Information on Grenada was gathered from a number of participants who freely shared their experiences with me. All of the interviewees provided crucial information but I

iv

especially appreciate the assistance of MAJ Dave Nolan and CPT Doug Phelps. Without their willingness to spend their time relating their experiences and observations, I could not have composed any real narrative about Grenada medical support.

To Dr. Robert Mosebar I owe special thanks for his help in assessing the medical support both in the Falklands and Grenada. His vast experience and insightful observations have been crucial in helping me deal with the problems of understanding the important medical lessons of each operation.

Ms. Judy Ronk, medical librarian at Munson Army Community Hospital, was extremely helpful in locating sources both on Falklands medical support and field medical support in general. She has great expertise in her field and provided me the finest support possible.

Of course, any errors in the telling of the stories of the medical support in the Falklands Campaign or the Grenada Expedition are wholly mine. I have sincerely tried to relate the events as they occurred and draw from them the salient conclusions and lessons which they provide for future medical support operations of rapidly deployed land forces.

Lastly, and most importantly, I would like to acknowledge the loving support of my wife, Beth, and my daughters, Karen and Susan. The number of hours invested in this thesis came from them, and I would not have been able to complete it without their help and encouragement.

v

## TABLE OF CONTENTS

		гаде
ABSTRACT		iii
ACKNOWLE	DGEMENTS	iv
TABLE OF	CONTENTS	vi
LIST OF	MAPS	viii
Chapter		
I.	INTRODUCTION	1
	Background: Rapid Projection of Land Forces	2
	Medical Support of Combat Operations of	-
	Rapidly Deployed Land Forces	5
	Statement of the Thesis Subject	7
	Objective <b>a</b>	7
	Assumptions	8
	Definitions	9
	Limitations	11
	Delimitations	11
	Significance of Study	12
	Outline of Subsequent Chapters	13
	Endnotes	15
II.	REVIEW OF LITERATURE	17
	Purpose	17
	Current Doctrine Publications	18
	Sources and Source Documents on the Falklands	10
		• •
	Campaign and the Grenada Expedition	23
	Methodology	26
	Endnotes	28
III.	MEDICAL SUPPORT PRINCIPLES AND CONCEPTS	29
	Combat Service Support Principles	30
	Medical Support Principles	32
	Medical Operational Concepts	34
	Triage	34
	Echelons of Medical Support	36
	Elements of Combat Medicine	38
	Patient Evacuation	40
	Command and Control	42
		43
	Endnotes	44

1

-----

IV.	MEDICAL SUPPORT IN THE FALKLANDS CAMPAIGN	47
	The Falklands Campaign: Background A Brief Review of the Combat Service Support	47
	of the Falklands Campaign	50
	Medical Support in the Falklands	55
	The Planning of the Medical Support	55
	The Execution of the Medical Support	60
	Command and Control of the Medical Support.	68
	Summary and Lessons Learned	70
	Endnotes	73
۷.	MEDICAL SUPPORT IN THE GRENADA EXPEDITION	81
	The Grenada Expedition: Background	81
	A Brief Review of the Combat Service Support	
	of the Grenada Expedition	84
	Medical Support in the Grenada Expedition	90
	The Planning of the Medical Support	90
	The Execution of the Medical Support	95
	Command and Control of the Medical Support.	102
	Summary and Lessons Learned	105
	Bndnotes	109
VI.	COMPARATIVE ANALYSIS	116
	Adherence to Principles	116
	Use of Operational Concepts	121
		125
	Endnotes	130
VII.	CONCLUSIONS AND RECOMMENDATIONS	131
	Conclusions	131
	Recommendations	132
	Recommendations for Further Study	135
	Endnotes	139
SELECTED	BIBLIOGRAPHY	140
INITIAL	DISTRIBUTION LIST	155

----

「日本のない」のないでは、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」ので、「日本のない」の

. :

## LIST OF MAPS

Page

FIGURE 1.	Distance from U.K. to Falkland Is	53
FIGURE 2.	The Falklands Campaign Map	54
FIGURE 3.	Distance from U.S. to Grenada	88
FIGURE 4.	The Grenada Expedition Map	89

L

I

#### CHAPTER I

#### INTRODUCTION

The United States Army has a proud tradition of providing the American soldier with the finest medical support possible. The Army Medical Department, the Army's organization tasked to provide medical support, has steadily improved the medical support of soldiers in every war America has fought. This fact is evident when the statistics since the Civil War are studied.

During the Civil War 17 percent of the soldiers who were wounded and reached a medical treatment facility died, while in World War I, this figure dropped to 8.1 percent and in World War II it was down to 4.5 percent.<sup>1</sup> In the Korean conflict only 2.5 percent of U.S. personnel who were wounded and reached a medical treatment facility died.<sup>2</sup> In Vietnam less than 1 percent of the battle-injured soldiers died after reaching an Army hospital.<sup>3</sup>

It is important to note that these percentages represent only the wounded personnel who died after reaching medical treatment facilities. In the Vietnamese conflict a far greater percentage of wounded personnel reached medical facilities alive than during any prior conflict. This was due primarily to the use of helicopter ambulances which were

able to rapidly evacuate severely wounded casualties who would have died in prior wars before the evacuee could have reached a medical treatment facility.4

Unquestionably the medical service in Vietnam was the most complete and timely medical support the American soldier has ever had. The medical support rendered there has clearly created expectations by which all future medical support will be judged. This certainly appears appropriate, but if the expectations are to be met, the Army Medical Department must closely analyze lessons learned from recent combat operations--operations which portend the increased use of ground combat forces in short duration employment: far from logistical bases.

#### Background: Rapid Projection of Land Forces

The United States has maintained forces capable of rapid deployment to potential contingencies throughout much of its history. But only recently, as the forces of the Soviet Union have expanded and the advances of technology have created a truly interdependent world community, have U.S. forces had to be prepared to deploy within hours to protect our vital interests and the vital interests of our allies.

The need to rapidly project land forces was forcefully brought to the American public's attention in 1979. This

came about through three events which began a dramatic reinflation of U.S. military aspirations and reversed the trend of the post-Vietnam era.<sup>5</sup> First, debate on the SALT II Treaty revealed the significant deterioration of U.S. military forces vis a vis Soviet forces.<sup>4</sup> Second, the overthrow of the Shah of Iran and the "seizure of U.S. diplomatic personnel as hostages in Tehran ... underscored American's vulnerability and helplessness."<sup>7</sup> Third, the Soviet Union invaded Afghanistan in December 1979.

In his January 20, 1980, State of the Union address, President Carter proclaimed the Carter Doctrine. In it he declared that any attempt by any outside force to gain control of the Persian Gulf region would be regarded as an assault on the vital interests of the United States. Such an assault, he said, would be repelled by any means necessary, including military force.<sup>4</sup>

For the first time, U.S. military forces were formally committed to defend Southwest Asia, a logistically remote region where the United States does not possess secure military access ashore in peacetime.<sup>9</sup> The Carter Doctrine thus "imposed new and exceedingly difficult obligations on U.S. conventional forces already severely overtaxed by traditional commitments in Europe and the Far East."<sup>10</sup>

The gap between the military aspirations espoused by President Carter and the forces available prompted action that has resulted in military force modernization initiatives during the past seven years. Included in these initiatives

have been enhancements of the U.S. Army's special operations forces--Rangers, Special Forces, special operations aviation units--as well as conventional forces.<sup>11</sup> Although force modernization has not been completed, the progress that has been made has increased the Army's capability to handle missions requiring the rapid deployment of ground combat forces.

Additionally, the use of military power has become more accepted as a means to deal with situations abroad especially when the safety of American citizens is involved. This is true not only because this is the view of the current Reagan Administration, but also because of the national will of the American people which grew out of the protracted American hostage crisis in Iran from 1979 to 1981.

Yet, as Colonel Harry G. Summers of the Strategic Studies Institute has observed, there have been no dramatic improvements in recent years in the ability of a nation to project power and fight a war far from home. Even with all of the high-technology weapons that are now available, the well-trained and capably led foot soldier is still the key to victory.<sup>14</sup> And, as the need for a rapidly deployable land force has been recognized and forces have been enhanced for rapid deployment missions, the possible uses of that land force have appeared to increase. The United Kingdom's campaign in the Falklands is one example of the successful use of rapidly deployed land forces. Another is the United States' Grenada Expedition.

4

ጜጜጜጜጚዾዄኇዀዀዿፙጜቘዾዄጜኯዀቘጟጚዹጞጜዀጜዀጚዀጚዀጚዀዄጞፙጚዀፙዿዀዀዀጚዀፙጜጚዀፙ፝ዿፙቚዀ

In view of the world situation today, rapidly deployed land force operations will be used increasingly. Rapidly inserting a force on the ground in a trouble spot appears to our national leaders to be the best way to handle a number of potentially explosive situations. Indeed, the advent of the U.S. Army's light infantry divisions and the expansion of special operations units grew out of this realization. One of the best ways to control volatile situations is to rapidly deploy superior and flexible combat forces so that opponents abandon their plans in reaction to U.S. maneuvers.<sup>13</sup>

Due to the increasing likelihood of more rapid deployment force operations, all military professionals must examine these operations. By doing so, all the lessons available from previous operations can be gleaned and future operations can be planned and executed with greater precision and higher chances for success.

# Medical Support of Combat Operations of Rapidly Deployed Land Forces

Since their purpose is to gain the initiative and impose a decisive force on the ground, the combat operations of rapidly deployed land forces must be accomplished with precision. Consequently, ground combat forces must plan

their missions carefully and execute them aggressively. The combat service support of such an operation must also be accomplished with precision. This is especially true with respect to medical support because ineffectiveness and inefficiency can cost lives.

The need for precision in medical support in rapidly deployed land force operations can hardly be overemphasized. As has already been shown, the expectations of medical care for soldiers wounded in combat operations are high. The tremendous record of medical support in the Vietnam conflict underscores the expectations of the U.S. Army Medical Department by the American public and today's soldier.

Supporting a rapidly deployed ground combat force can pose unique problems in providing timely medical support. This is because of the shortage of time to plan, the difficulty of forecasting the locale of employment, and the great distance at which medical support may have to be provided away from large medical support complexes. For these reasons, the medical support of combat operations like the Falklands Campaign and the Grenada Expedition must be studied to determine if established medical support principles were followed and what the actual execution of the medical support can teach us to improve future operations.

#### Statement of the Thesis Subject

To compare American and British medical support of combat operations of rapidly deployed land forces. The researcher will conduct a comparative analysis of the medical support of the British Army in the Falklands Campaign and the U.S. Army in the Grenada Expedition.

#### **Objectives**

The objectives of this study are: (1) to determine what the medical support principles are for U.S. Army and British Army combat operations; (2) to determine if the principles were followed in the Falklands and Grenada respectively; and (3) to draw conclusions from a comparative analysis of both operations.

The medical support principles will be examined in the context of the respective army combat service support principles. The medical support principles of both armies will also be compared to each other.

Medical support principles will be determined by a review of the published doctrinal manuals for each army. Medical support principles and actual medical support operations will be examined at corps level or lower. The principles established by the U.S. Army Medical Department and the Royal Army Medical Corps will then be

compared and contrasted with each other to determine if they provide an adequate framework for planning and executing medical support of combat operations.

#### Assumptions

This study assumes that the requirement for medical support of combat operations is valid and that the expectations of medical support of combat operations, at least for the U.S. Army, are the expectations based on the Vietnam experience. Research and analysis will start from these points.

That medical support is one aspect of combat service support is a fact and medical support must, therefore, be examined in the context of the overall combat service support plan of any operation. Medical support, both in planning and execution, is unique, however, and must be reviewed in the context of special medical principles and concepts.

Planners and executors of rapidly deployed land force operations should be concerned with appropriate medical support of those operations. Like all military operations, rapid deployment force operations require adequate combat service support both to achieve initial success and to sustain that success for the duration of the operation. Therefore, the medical support which sustains the soldiers in the operation is important and is a key element in the

force's ability to complete its mission. Medical support is as essential as any part of the combat service support equation.

Actual combat operations studied are the Falklands Campaign and the Grenada Expedition. Medical support principles and operational concepts are those described in current U.S. Army and British Army field operations manuals. Primarily referenced will be U.S. Army Field Manuals 8-10 "Health Service Support in a Theater of Operations" and 8-55 "Planning for Health Service Support" and the British Army's "Administration in War" and "Medical Interoperability Handbook."

#### <u>Definitions</u>

This study will use operational terms peculiar to U.S. Army medical support as well as terms peculiar to British Army medical support. All terms will be defined in accordance with respective army doctrinal publications and British Army terms will be carefully related to the most similar U.S. Army term.

For the purpose of this study, a rapidly deployed land force will be defined as a tailored military package rapidly assembled and deployed to meet an urgent situation. Such a force is composed of essentially standard military units but is assembled for a unique mission.

To serve as a framework for conducting the comparative

analysis which is the focus of this thesis, certain principles, doctrines, and operational concepts will be identified and discussed. In the context of this research, the words principle, doctrine, and operational concept are defined in the following manner:

 (1) principle - a fundamental truth used to guide the planning and execution of a military operation which is sanctioned in official military publications;

(2) doctrine - a body of fundamental truths or principles sanctioned in official military publications;

(3) operational concept - an idea about the conduct of military operations formed by generalization from historical military experiences and sanctioned in official military publications.

In this study, field medical support consists of unit level, division level, and corps level medical support. Although there may be higher levels of medical support deployed with the force, such medical support will be discussed only as it relates to the echelons of medical support at corps or below.

Unit level or first echelon medical support is primarily concerned with the provision of emergency medical treatment and evacuation of the wounded from battle areas as necessary. First echelon medical support is provided by unit aidmen or by battalion aid post personnel.

Division level or second echelon medical support is provided by medical companies of division medical battalions

or forward support battalions. A medical company, normally placed in support of a brigade-size force, provides definitive treatment for relatively minor injuries or resuscitates and stabilizes casualties with more complex injuries for evacuation to a corps level hospital for initial surgery.

#### Limitations

This study will not address non-medical aspects of combat service support of the two actual operations in detail. Research will focus on the planning, execution, command and control of the medical support operations.

Additionally, this study will concentrate on medical support of combat operations and not concern itself with medical support once combat operations in Grenada and the Falklands had ceased.

#### **Delimitations**

This study will not examine the political considerations in the employment of forces in Grenada, in the Falklands, or elsewhere. The focus of the study is a comparative analysis of the medical support of the combat operations in Grenada and in the Falklands and how the medical support in each

operation compares with common medical support principles.

Historical research will be limited to combat operations since the end of the Second World War. Although this study focuses on medical support of combat operations in a low and mid-intensity environment, it will include medical support considerations across the whole spectrum of battlefield intensity.

Since land forces may have to be rapidly deployed worldwide, this study will not be limited geographically. The medical support principles examined will be applicable to any area in the world where combat operations may take place. Consequently, different types of climatic areas in the world will not be specifically addressed.

#### Significance of the Study

As mentioned earlier in this introductory chapter, the lessons of medical support of recent combat operations must be closely reviewed. This is necessary to ensure that medical support of future rapid deployment force operations is conducted with the benefit of an analysis of those lessons. The present recognition of the significance of rapidly applied military force in the international arena clearly points out that rapid deployment force operations will see more extensive use in the future. To the U.S. Army Medical Department and the soldiers it serves, the

effectiveness and efficiency of combat medical support are crucial.

### Outline of Subsequent Chapters

.

Chapter Two reviews the available literature which bears on a comparative analysis of the medical support of the combat operations in Grenada and the Falklands. Doctrinal publications by the U.S. Army and the British Army are reviewed to assess their adequacy for the planning and execution of medical support of rapidly deployed land forces. From this review principles will be established in the subsequent chapter to evaluate the actual conduct of the medical support operations. Publications on the actual combat operations are also reviewed with emphasis on the publications about the medical support of each operation.

Chapter Three specifies the relevant principles for medical support as well as for overall combat service support. Principles are identified and discussed for both the U.S. Army and the British Army. The principles and operational concepts of each army are also compared to point out similarities and differences.

Chapter Four examines the medical support provided during the Falklands Campaign and assesses whether established medical support principles and concepts were or were not followed.

Chapter Five analyzes the medical support provided to American forces in the Grenada Expedition. It also assesses whether established medical support principles and concepts were or were not followed.

Chapter Six compares the medical support in the Grenada Expedition with the medical support in the Falklands Campaign in the context of the commonly shared medical support principles and concepts.

Chapter Seven provides conclusions and recommendations. Recommendations are made as to the principles and operational concepts needed to insure future successful medical support of combat operations by rapid deployment forces. Suggestions are also given to guide future researchers.

14

#### ENDNOTES

#### CHAPTER I

LTG Joseph M. Heiser, Jr., <u>Vietnam Studies: Logistics</u>
 <u>Support</u> (Washington, D.C.: Department of the Army, 1974),
 p. 212.

2. MG Spurgeon Neel, <u>Vietnam Studies: Medical Support of the</u> <u>U.S. Army in Vietnam 1965-1970</u> (Washington, D.C.: Department of the Army, 1973), p. 51.

3. "Copter Ambulances Improve Survival Rate," <u>The Journal of</u> the <u>Armed Forces</u> 103 (December 11, 1965): 12.

4. Heiser, Logistics Support, p. 213.

5. <u>Strategic Studies: National Security Policy Considera-</u> <u>tions</u> (Fort Leavenworth, Kansas: U.S. Army Command and General Staff College, 1986), p. 159.

6. Ibid.

7. Ibid.

8. President Jimmy Carter, A Report on the State of the

Union 96th Congress House Document No. 96-257 (Washington, D.C.: U.S. Government Printing Office, January 22, 1980), p. 4.

9. Strategic Studies, p. 160.

10. Ibid.

11. COL James B. Motley, "Wathington's Big Tug of War Over Special Operations Forces," <u>Army</u> 36 (October 1986): 18.

12. COL Harry G. Summers, "Yomping to Port Stanley," <u>Military</u> <u>Review</u> 64 (March 1984): 3.

13. Congressman Richard B. Cheney and MAJ Thomas N. Hervey, "Strategic Underpinnings of a Future Force," <u>Military Review</u> 66 (October 1986): 10.

#### CHAPTER II

#### REVIEW OF LITERATURE AND METHODOLOGY

#### Purpose

This chapter provides a review of the literature on medical support of combat operations relevant to this thesis. It serves as a point of departure for the research presented in the subsequent chapters and shows the foundation upon which the research analysis is based. The chapter is organized into three main sections.

The first section deals with publications on current medical doctrine, but not the doctrine itself. The actual doctrine of both armies is discussed in detail in Chapter Three. This section presents information on the doctrinal publications of both the American Army and the British Army. Combat service support doctri detail support doctrine are addressed by these publications and they are the source

of the doctrine upon which the actual conduct of medical support operations is based. A brief assessment of the adequacy of the doctrine provided by these manuals is made. Also a brief comparison of the doctrinal publications of the U.S. and British Armies is conducted. Chapter Three presents a more detailed comparison.

The second section of this chapter discusses the sources available on the actual conduct of both of the operations under study. Source documents on the Grenada Expedition and the Falklands Campaign are reviewed and information on the techniques of searching for research material is presented. A review of the availability and use of unclassified publications on the medical support of Grenada and the Falklands is addressed. Additionally, the availability of classified sources is discussed.

The third section, entitled Methodology, delineates the method for conducting the comparative analysis of the medical support doctrines. It also specifies how the comparative analysis of the actual medical support operations in the Grenada Expedition and the Falklands Campaign is conducted.

#### Current Doctrine Publications

The current doctrinal information on medical support of combat operations is extensive in the U.S. Army and is found

in Army field manuals (FMs) and field circulars (FCs). General concepts of medical support are addressed in Army publications on combat service support (CSS) operations. Detailed doctrinal information on medical support is contained in the 8 - series of Army field manuals and circulars. CSS doctrine publications are reviewed first because this resourcher believes that medical support must be examined in the context of combat service support operations.

FM 100-10 "Combat Service Support" is the U.S. Army's keystone manual on the principles of combat service support of forces committed to battle. It is the lead volume of the Army's series of "how-to-support" field manuals and presents general concepts developed in greater detail in the how-tosupport FMs.<sup>1</sup> The other field manuals in the series are:

FM 63-1 "CSS Operations - Separate Brigade"
FM 63-2 "CSS Operations - Division"
FM 63-3J "CSS Operations - Corps"
FM 63· 'SS Operations - Theatre Army Area Command"

FM 63-5 "CSS Operations - Theatre Army"

These manuals provide detailed doctrinal information on the conduct of CSS operations at the echelons designated above. For this thesis, FMs 63-2, 63-3J, and 100-10 are the most relevant. These are adequate publications which are concise and clearly written.

An additional doctrinal publication which has proved valuable in researching CSS principles has been the U.S. Army Command and General Staff College's summarization of the

principles contained in FM 100-10 and the FM 63-series. This summarization is found in <u>Introduction to Combat Service</u> <u>Support</u> used in the college's curriculum. <u>Introduction to</u> <u>Combat Service Support</u> synthesizes eight broad principles of CSS highly relevant to this research.<sup>2</sup>

Like FM 100-10, "Administration in War" (British Army Code Number 71342) is the British Army's keystone combat service support manual. The term administration is defined by this British Army manual as "the management and execution of all military matters not included in tactics and strategy; primarily in the fields of logistics and personnel management."<sup>3</sup> "Administration in War" is similar to FM 100-10, but it is a much more detailed manual. It not only presents basic principles like 100-10 but also provides details about different echelons much as the U.S. Army's 63series of how-to-support manuals. This British Army manual and several others were obtained from the U.S. Army Command and General Staff College's British Liaison Officer, COL R.J.B. Heard.

The British Army supplements their primary doctrinal manuals such as "Administration in War" with precis, or brief summaries covering new developments or elaborations of information on a particular subject. For example, one of the precis used in this thesis was "Logistics 2" which outlines administration and logistics in the division in war and is used in the British Army Staff College.4

As mentioned in the beginning of this section, U.S. Army

8-series field manuals and circulars deal with medical support of combat operations. These manuals and circulars are essentially "how-to-medically-support" publications. Although they are not as consistent in dealing with progressive echelons of support like the 63-series they do provide detailed information in a highly readable and useful format. FM 8-10 "Kealth Service Support in a Theater of Operations" clearly specifies essential medical support principles while FM 8-55 "Planning for Health Service Support" comprehensively addresses all aspects of planning medical support of combat operations. Also significant is FM 8-8 which deals with "Medical Support in Joint Operations." Currently under revision, it provides guidance about Army, Navy, and Air Force medical support in an area of operations as well as joint medical planning factors and procedures.

Recently the Army has published several highly relevant and excellent medical support field circulars. These are FC 8-15-1 "Health Service Support Operations-Light Infantry Division," FC 8-15-2 "Health Service Support Operations-Airborne and Air Assault Divisions," and FC 8-45 "Medical Evacuation in the Combat Zone." All three address support of combat operations and represent the most current conceptual publications of the Army Medical Department related to this study. Also highly relevant is TRADOC Pamphlet 525-50 "Health Service Support AirLand Battle."

Essentially the same medical support concepts outlined in the U.S. Army doctrinal publications are contained in the

British Army's "Administration in War." The specific similarities and differences between U.S. and British Army doctrines will be discussed in the next chapter. "Administration in War" clearly presents the role and organization of British Army medical support systems. Medical support is further addressed in the "Medical Interoperability Handbock" (British Army Code Number 71376) and in a medical precis entitled "Medical Support in the Field"--both provided by Colonel Ian Creamer, Royal Army Medical Corps liaison.

Few publications in either military or medical journals deal with changing current military medical doctrine. A search of both American and British journals reveals a scarcity of supplemental publications either debating the validity of current dostrine or setting forth recommendations for change. This researcher believes this is due in part because few journals attract articles on medical doctrine and because there are relatively few proponents for change willing to prepare such articles. The primary reason that few supplemental publications on medical doctrine are found is because current doctrine has developed from time-tested principles and concepts. These principles and concepts have been derived through the experience of both American and British medics in the wars each country has experienced in the 19th and 20th Centuries. The doctrine is sound. Any debate grows out of how closely the doctrine is followed in support of actual combat operations. Generally speaking,

both U.S. and British Army medical support doctrines are clearly written and conform to the tactical doctrines they support.

# Sources and Source Documents on the Falklands Campaign and the Grenada Expedition

Since the focus of this thesis is a comparative analysis of the medical support of two operations by rapidly deployed land forces, the literature review next concentrated on the availability of material on the Falklands and Grenada. Several computer searches were conducted via both military and medical channels to obtain source material. This material was examined on three levels: (1) general information on each operation; (2) information on the combat service support of each operation; and (3) specific information on the actual medical support of the combat operations in the Falklands and Grenada.

Information on the first level was extensive for the Falklands Campaign both through official British Army sources and unofficial sources. Many books have been written about the Falklands Campaign. This researcher confined himself to the general references which concentrated on ground combat operations. A number are cited in the bibliography but the best are Hastings and Jenkins' <u>The Battle for the Falklands</u> and Frost's <u>2 PARA Falklands: The Battalion at War</u>. The

former is the most readable and historically accurate of the books dealing with the campaign as a whole while the account of the 2nd Parachute Battalion, by the Arnhem veteran John Frost, captures the ferocity of the Falklands battles.

With respect to Grenada, the only extensive sources available are official U.S. Army after action reports and assessments. These include: (1) Bishop and O'Brien's FORSCOM/ARLANT Participation in Operation Urgent Eury ---Grenada, 1983; (2) the Grenada Work Group's Operation Urgent Eury Assessment; (3) the Department of the Army's Lessons Learned Grenada: U.S.Army Lessons Learned From 1983 Operation Urgent Eury; and (4) Pirnie's Operation Urgent Eury: The United States Army in Joint Operations. Books on the Grenada intervention are scarce and much of the Army's material is classified. However, enough material in a number of unofficial sources is available to clearly reconstruct the essential events in the Grenada Expedition for this thesis.

With respect to sources on combat service support, the material on the Falklands also significantly exceeds that available on Grenada. CSS for the Falklands Campaign is fully addressed in a number of official after action reports as well as unofficial articles noted in the bibliography. A superb distillation of these articles and others was recently made by an officer in the U.S. Army Command and General Staff College's School of Advanced Military Studies. Major Kenneth L. Privratsky has written <u>British Combat</u> <u>Service Support on East Falkland: Considerations for</u>

<u>Sustaining Tactical Operations in Remote Areas</u> which was published in November 1985. This work has served as an extremely useful guide to overall CSS operations for the British campaign.

Although unofficial sources on Grenada combat service support were extremely limited, the four official after action documents previously mentioned did cover CSS in sufficient detail to effectively conduct this research.

Finally, with respect to the specific information on the actual medical support of the combat operations in Grenada and the Falklands, the same trend continued. Extensive official and unofficial sources are available on the Falklands Campaign. The USACGSC's British Liaison Officer, COL R.J.B. Heard, put this researcher in contact with the Royal Army Medical Corps' Liaison Officer who provided a wealth of official source material. In addition, more than a dozen articles in British military and medical journals were obtained which were written on the Falklands medical experiences. These articles covered the full range of medical support activities.

Information on medical support in Grenada was limited to two types of sources. These were the official after action reports and interviews with medical officers who participated in the Grenada Expedition medical support. The combination of the information derived from both sources gave the researcher a detailed picture of the medical support provided.

Compared to Grenada, little material on the Falklands is classified. Although the principal sources about the Grenada Expedition are classified there is abundant unclassified material contained within the reports which has been used in this thesis. There was no need to use classified material to research the Falklands Campaign and the obstacle of classified documents on Grenada was overcome because of the availability of unclassified sections within the classified reports.

Finally, a number of sources (both books and periodicals) on combat service support and medical support were consulted for historical perspective. The best of these was Major General Spurgeon Neel's <u>Medical Support of the U.S.</u> <u>Army in Vietnam 1965-1970</u>, a thorough recounting of the Army Medical Department's support in the Vietnamese conflict. The other prominent historical analysis was <u>Rapid Deployment</u> <u>Logistics: Lebanon 1958</u> by Lieutenant Colonel Gary H. Wade. As Major General Dave L. Palmer states in the foreward to that work, "a good number of the logistical problems encountered in Lebanon in 1958 recurred in the U.S. intervention in Grenada twenty-five years later."<sup>\$</sup>

#### <u>Methodology</u>

This investigation is founded upon a thorough understanding and a concise explanation of the principles of medical support of combat operations. These principles are discussed in Chapter Three and have been examined from the viewpoints of both the American and British Armies. They have also been placed in the context of the principles for overall combat service support of each army. This has been essential to clearly delineate the role of medical support in relation to other components of combat operations service support.

Planning and execution of the actual CSS operations and medical support operations are reviewed as well as the aspect of command and control. As in all military operations, planning will be shown to be vital to effective execution of combat service support in the Falklands and Grenada. The planning processes of both armies have been examined per their doctrine and in the actual situations in the two operations. Execution of support operations has been analyzed to determine how closely the actual support followed plans and how effective the actual support was judged to be by participants and expert observers.

To reiterate, the focus of this thesis is a comparative analysis of the medical support of the combat operations in the Grenada Expedition and the Falklands Campaign. Doctrine of both armies is compared. Evidence collected is analyzed with respect to the established medical support principles as derived from U.S. Army and British Army field manuals. From this analysis conclusions are derived and presented.
#### ENDNOTES

### CHAPTER II

1. U.S. Department of the Army Field Manual 100-10 "Combat Service Support." : ii.

2. LTC John R. Raffle, <u>Combat Service Support</u>: <u>Introduction</u> to <u>Combat Service Support S480/6</u>. U.S. Army Command and Staff College, Fort Leavenworth, Kansas, 13 May 1985: 19-21.

3. "Administration in War." British Army Manual on Administration and Logistics. Army Code Number 71342. Ministry of Defence, 1984: 1-1.

4. "Logistics 2." Precis from the United Kingdom Army Staff College in files of British Liaison Officer, U.S. Army Command and Staff College.

5. LTC Gary H. Wade, <u>Rapid Deployment Logistics: Lebanon</u> <u>1958</u>. Combat Studies Institute, U.S. Army Command and Staff College, Fort Leavenworth, Kansas, October 1984: foreword.

## CHAPTER III

# MEDICAL SUPPORT PRINCIPLES AND CONCEPTS

To understand and analyze the medical support of combat operations of the American and British Armies, one must first examine the essential principles and operational concepts upon which that medical support is based. These concepts and principles are stated clearly in the medical support doctrinal publications of both the U.S. Army Medical Department (USAMEDD) and the Royal Army Medical Corps (RAMC). They grew out of and are grounded in the principles and concepts of combat service support of both armies and special medical considerations. Consequently, a brief review of combat service support tenets for both armies is presented first. Then medical support principles and concepts will be examined in depth.

### Combat Service Support Principles

The American Army defines combat service support (CSS) as "the assistance provided to sustain combat forces, primarily in the fields of administration and logistics."<sup>1</sup> Administration refers to personnel service support (including medical support) and civil affairs administration. Logistics includes maintenance, transportation, facilities, and supply (excluding medical supply).

The British Army uses the word "administration" instead of combat service support and candidly points out in "Administration in War" that the terms "administration" and "logistics" are frequently used loosely.<sup>2</sup> Except for this point, definitions of "combat service support" in the American Army and "administration" in the British Army are essentially identical. In fact, since both nations are members of the North Atlantic Treaty Organization (NATO), both define "logistics" as planning and carrying cut the movement and maintenance of forces - the standard NATO definition.<sup>3</sup>

The British Army identifies five principles of administration: foresight, economy, flexibility, simplicity, and cooperation.<sup>4</sup> These principles are augmented by the principle that "forward planning will be essential to the success of an operation since resupply may be complex and the maximum use must be made of local resources in cooperation

with a host nation or ally."<sup>5</sup> The British also emphasize, in a more general sense, that "no plan can be formulated without an accurate appreciation of the logistic intelligence available and of all other relevant factors."<sup>6</sup>

Similarly the American Army incorporates "the principles of responsiveness, flexibility, and initiative" in their combat service support doctrine.<sup>7</sup> These are expanded to form the following eight principles which are identified as the basis of U.S. Army CSS doctrine:

- (1) Support Forward
- (2) Continuous Planning
- (3) Practice Economy
- (4) Maximum Use of Standing Operating Procedures
- (5) Tailored Support
- (6) Centralized Control/Decentralized Operations
- (7) Maximum Use of Throughput Distribution
- (8) Maximum Use of Local Resources \*

From this brief discussion one can clearly see that both armies base their service support of combat operations on highly similar principles. Differences are minor and are ones of language rather than substance. The eight principles listed above not only form the basis of American Army CSS doctrine but can also be used to examine British Army administration in war. They form a common context within which medical support principles are established and can be

examined for comparing American and British medical support operations.

### Medical Support Principles

Unlike the combat service support principles which were briefly reviewed, the following medical support principles and concepts of the American and British Armies will be more fully examined and discussed. Certainly medical support principles and operational concepts are central to the comparative analysis which is the focus of this thesis.

To provide effective and efficient medical support of combat operations, the USAMEDD and the RAMC have established a number of general principles. The American Army specifies these in FM 8-10 and enumerates six principles.\* These are:

- (1) Conformity
- (2) Proximity
- (3) Flexibility
- (4) Mobility
- (5) Continuity
- (6) Control

Conformity points out that medical support must conform to the tactical plan of operation so that medical support is

provided at the right place and time. Proximity stresses that medical support must be provided as close to combat operations as possible to minimize morbidity and mortality. Flexibility emphasizes that units providing medical support must be prepared to shift medical support resources to meet changes in tactical operations.

Mobility is an essential principle because medical support units must maintain close contact with maneuvering tactical units. Continuity provides for moving the patient through a progressive and phased medical support system in an uninterrupted manner that decreases morbidity and mortality. Finally control stresses the need for centralized management of medical resources to maximize the treatment medical support units can offer to soldiers in combat operations.

Similarly, the British Army identifies basic principles for the collection, evacuation and treatment of their wounded soldiers in "Administration in War."<sup>10</sup> The British emphasize tactical congruency, close support, adaptability, and the collection of casualties as quickly as possible (conformity, proximity, flexibility, and mobility). They also stress the importance of continuity of care and control of medical resources. As in the CSS principles, minor differences in language exist between British and American medical support principles. Nevertheless, the general principles for medical support are shared in common.

# Medical Operational Concepts

Medical operational concepts of the USAMEDD and the RAMC are also highly similar. A comparison of operational concepts, however, points cut the first significant differences between American and British medical support. The medical operational concepts of the USAMEDD and RAMC encompass the following: (1) triage; (2) echelons of medical support; (3) elements of combat medicine; (4) patient evacuation; and (5) command and control.

### Trigge

Both the British and American Armies use the concept of triage (or patient sorting). Triage, which in French means "the division into three," began as a concept to separate patients into three groups. One group needed immediate attention, one group could wait, and one group had wounds so severe that life could not be saved due to insufficient time, medical personnel or resources.

Triage is an essential concept for handling large numbers of casualties (called mass casualty situations). This concept applies to all medical support of combat operations because of the ever-present likelihood of mass casualty situations arising.

Both the American and British Armies use the NATO definition of a mass casualty situation and the four (not three) NATO treatment group categories. NATO defines a mass casualty situation as one in which an overwhelming number of seriously injured are placed upon medical facilities unable to supply medical care for all. "Under these conditions," the definition states, "the aim must be to assure care to the greatest benefit of the largest number."<sup>11</sup> The USAMEDD and the RAMC use the four general treatment categories below:

- Immediate for patients requiring immediate treatment to save life or limb;
- (2) Delayed ~ for patients who, after emergency treatment, can have major definitive
   , procedures delayed;
- (3) Minimal for patients who need simple treatment
  and can be returned to duty immediately;
- (4) Expectant for patients with messive injuries and little chance of survival even if all medical resources are concentrated upon them.<sup>12</sup>

The use of NATO terminology with respect to the operational concept of triage thus eliminates any significant difference between the USAMEDD and the RAMC.

#### Echelons of Medical Support

Both the American Army and the British Army recognize the necessity for establishing CSS concurrently with the employment of tactical units. FM 100-10 states that "some combat service support elements should be employed into a hostile environment as soon as the first forces land."<sup>13</sup> This objective is achieved by both armies by echeloning medical support.

The first echelon of medical support in the U.S. Army is unit level medical support. This is provided by medical aidmen and aid stations organic to tactical units. In the British Army, unit level medical support is called first line medical support and is essentially the same as for the American Army.<sup>14</sup> An important supplement to this level is buddy-aid. Both armies train their soldiers in basic lifesaving measures and first-aid treatment. This enables a soldier to administer to himself or to a fellow soldier if the unit medic cannot immediately attend to him.

The second echelon of USAMEDD support is division level medical support. Division level medical support "includes evacuation of patients from unit level aid stations and initial resuscitative treatment."<sup>15</sup> This support is provided by medical ompanies and sections of the division medical battalion or forward support battalions. The equivalent of division level medical support in the British Army is called

second line medical support. The major second line unit is the field ambulance whose tasks are: (1) collection of casualties from unit (usually regimental) aid posts; (2) centralization of casualties at dressing stations (like American medical companies); (3) casualty treatment "to enable them to survive evacuation to hospitals or return to duty;" and (4) control of patient evacuation.<sup>16</sup>

Hospitals are first found in the third echelon of each army and their role at this echelon is the same for both armies. "This level of support includes the evacuation of patients from supported divisional and non-divisional units, resuscitative surgery and emergency/resuscitation care on an area basis....<sup>17</sup> This is also the first level where surgery is normally performed. USAMEDD field medical units in the third echelon include 60-bed mobile army surgical hospitals (MASHs), 200-bed combat support hospitals (CSHs), and 400-bed evacuation hospitals (EVACs).<sup>18</sup>

Third echelon RAMC units include 400-bed field hospitals (Forward Support Complexes) and 800-bed general hospitals (Rear Surgical Complexes).<sup>19</sup> Both RAMC and USAMEDD hospitals at this echelon may detach small surgical teams down to second echelon medical units or position the teams as lead or advance elements of the third echelon medical support. This practice is found in both armies for medical support in remote areas with slow means of patient evacuation. By doctrine, third echelon hospitals are the highest echelon established by both armies in the combat zone.

The highest echelon of medical care for both armies is the fourth echelon. In addition to having the capability of providing all medical treatment available at the other echelons, fourth echelon hospitals can provide definitive care for all patients.<sup>20</sup> These hospitals are usually found in the communications zone behind or out of the area of combat operations. They may be permanent facilities or they may occupy semi-mobile, tent-sheltered facilities. They include fixed medical centers operated by both the USAMEDD and the RAMC. USAMEDD hospitals at this echelon also include field hospitals and general hospitals while the RAMC hospitals are 400-bed evacuation hospitals and 800-bed general hospitals.<sup>21</sup>

Due to their size and the wide range of medical care which they provide, third and fourth echelon hospitals are augmented by large evacuation units (evacuation battalions or ambulance regiments) for evacuating patients from the echelons of medical support immediately below them. In both the American and British armies, first and second echelon medical units have organic evacuation assets. (A subsequent section of this chapter discusses patient evacuation.)

### Elements of Combat Medicine

The type of treatment available for patients at the third and fourth echelons of military medical care is

essentially like that found in any civilian hospital or medical center. The type of treatment found at the first two echelons of medical support in both the American and British Armies requires more specific explanation, however, because the type of medical care given is directly related to the fact that the care is given in a tactical environment.

The USAMEDD structures medical care at the unit-level and the division-level around the elements of advanced trauma life support (ATLS) care. These elements are:

- (1) assessment of patients with multiple injuries;
- (2) insertion of breathing tubes;
- (3) prevention and treatment of shock;
- (4) replacement of lost body fluids;
- (5) emergency treatment of trauma injuries; and
- (6) initial treatment of burns.

The USAMEDD's goal is to provide soldiers ATLS care within thirty minutes of injury and to stabilize patients requiring third and fourth echelon care so that surgical intervention can be provided within four to six hours. As can be seen, ATLS care is relatively basic medical care. Given the environment of combat operations and the inherent difficulties of providing medical support in that environment, ATLS care is the right level of care which the first and second echelons of medical support should provide.

The RAMC uses basically the same framework at their

first two echelons of medical support. Like the USAMEDD, the RAMC fully recognizes the criticality of providing soldiers lifesaving care as soon as possible. The timeframes they recognize for effectively providing that care are the same as the ones used in the American Army. In specifying the equipment which their personnel need in order to provide ATLS care, the British specify those medical items needed to encompass the six elements of combat medicine identified by the USAMEDD.<sup>22</sup>

# Patient Evacuation

A key concept of medical support is patient evacuation. The U.S. Army defines patient evacuation as the timely, efficient movement of wounded from the battlefield which begins where the injury occurs and continues until the patient receives the definitive care he needs.<sup>22</sup> The U.S. operational concept is that the gaining medical unit is responsible for patient evacuation - higher evacuates lower. The British Army operates its patient evacuation system in the same manner except for one significant difference.<sup>24</sup>

The significant difference concerns aeromedical evacuation. Although the USAMEDD relies heavily on ground ambulances, the USAMEDD specifies that the preferred means of patient evacuation is by air. "Aeromedical evacuation, to the maximum extent feasible, will be used in the combat zone

for evacuation."<sup>23</sup> The RAMC uses aeromedical evacuation also but unlike the USAMEDD the RAMC has no aeromedical evacuation units of its own. The RAMC must rely on CSS helicopters to perform aeromedevac (without in-flight medical care) in addition to their primary mission of moving equipment, supplies, and troops in airmobile operations.

Evacuation is a major element in the medical support of combat operations. The speed of evacuation and the continuous enroute medical care from the battlefield to the treatment facility are just as important as the emergency medical treatment at the site of injury and the comprehensiveness of medical care available at hospitals. As Major General Spurgeon Neel, U.S. Military Assistance Command Vietnam surgeon, observed about medical support in Vietnam, "Getting the casualty and the physician together as soon as possible is the keystone of the practice of combat medicine. The helicopter achieved this goal as never

The Korean and Vietnam experiences proved to the American Army that the USAMEDD should have its own helicopters for aeromedevac. According to the U.S. Army, air ambulances are as essential as ground ambulances. They provide a new dimension for patient evacuation and their control by medical commanders rather than general CSS commanders is as appropriate as is the control of ground ambulances. To date, the RAMC has not persuaded the British Army to a similar point of view.

## Command and Control

A final operational concept which is central to any comparison of medical support of combat operations is command and control. A basic understanding of command and control is key to an understanding of any military operation.

Paraphrasing FM 101-5-1, command and control in the medical support arena is the process of directing, coordinating and controlling medical units to accomplish the medical support mission. "The process encompasses the personnel, equipment, communications,...and procedures necessary to gather and analyze information...," plan medical support operations and supervise their execution.<sup>27</sup> Unity of command is recognized as essential for the USAMEDD as it is for the other elements of the American Army. The role of the medical commander - at each echelon of medical support - is crucial to effective medical support of combat operations.

In the same way as in the American Army, medical command and control is recognized as absolutely vital in the British Army. At each of the levels of medical support which the RAMC provides, medical officers control the provision of patient treatment, the order of evacuation, and the network for medical communications.<sup>28</sup> The RAMC's "Medical Support in the Field" clearly describes command and control imperatives for every echelon. It carefully addresses the medical commander's responsibilities for resource allocation, medical

planning, information gathering, and casualty evacuation (casevac) coordination.<sup>29</sup>

The medical support of rapidly deployed land force operations requires the superior command and control described in the U.S. Army's keystone warfighting manual, FM 100-5 "Operations." Essential for this is the thorough understanding by all elements of a force of the overall commander's intent and concept of operations.<sup>30</sup> Both the USAMEDD and the RAMC fully recognize the significance of this understanding and emphasize its importance throughout their medical support doctrinal publications.

#### Summary

The principles and operational concepts discussed in this chapter provide an excellent framework for comparing the planning and execution of medical support operations. The following examination of the medical support in the Falklands and in Grenada uses this framework to analyze the planning, execution, command and control of those medical support operations.

#### ENDNOTES

### CHAPTER III

1. U.S. Department of the Army Field Manual 101-5-1 "Operational Terms and Symbols." : 1-16.

2. "Administration in War." British Army Manual on Administration and Logistics. Army Code Number 71342. Ministry of Defence, 1984: 1-1.

3. FM 101-5-1, p.1-43 and "Administration in War," p. 1-1.

4. "Administration in War," p. 1-2.

5. (bid., p. 1-7.

6. Ibid., p. 1-8.

7. U.S. Department of the Army Field Manual 63-2 "Combat Service Support Operations - Division." : 1-2.

8. LTC John R. Raffle, <u>Combat Service Support</u>: <u>Introduction</u> <u>to Combat Service Support S480/6</u>. U.S. Army Command and Staff College, Fort Leavenworth, Kansas, 13 May 1985: 19-21.

9. U.S. Department of the Army Field Manual 8-10 "Health Service Support in a Theater of Operations." : 2-12 and 2-13.

10. "Administration in War," pp. 2-54 and 2-60.

11. "Medical Support in the Field." Precis from the Royal Army Medical Corps Training Centre in files of British Liaison Officer, U.S. Army Surgeon General's Office: 1A-1.

12. FM 8-10, p. 6-3 and "Medical Support in the Field," p. 1A-1.

13. FM 100-10, p. 2-24.

14. "Medical Interoperability Handbook." British Army Manual Code Number 71376. Ministry of Defence, February 1986: 1-2.

15. FM 8-10, p. 2-4.

16. "Medical Interoperability Handbook," p. 1-3.

17. Ibid.

18. FM 8-20, pp. 4-1 thru 4-5.

19. "Medical Interoperability Handbook," p. 1-4.

20. U.S. Department of the Army Field Manual 100-16 "Support Operations: Echelons Above Corps." : 8-2.

21. "Medical Interoperability Handbook," p. 1-4.

22. "Medical Support in the Field " pp. 1-3 and 2C-5.

23. U.S. Department of the Army Field Marual 8-35 "Evacuation of the Sick and Wounded." : 1-1.

24. "Medical Interoperability Handbook," p. 4-10.

25. U.S. Department of the Army Field Manual 63-3J "Combat Service Support Operations - Corps." : 10-8.

26. MG Spurgeon Neel, <u>Vietnam Studies: Medical Support of</u> <u>the U.S. Army in Vietnam 1965-1970</u>. Washington, D.C.: Department of the Army, 1973: 59.

27. FM 101-5-1, pp. 1-16 and 1-17.

28. "Administration in War," p. 4-11.

29. "Medical Support in the Field," pp. 3-8 thru 3-11.

30. U.S. Department of the Army Field Manual 100-5 "Operations." : 3-4 and 21-21.

#### CHAPTER IV

# MEDICAL SUPPORT IN THE FALKLANDS CAMPAIGN

## The Falklands Campaign: Background

On the second of April, 1982, Argentina invaded the British-owned Falkland Islands. The Falklands, located 450 miles from Argentina and 8,000 miles from Britain, had long been claimed by Argentina. Prior to the invasion, Argentina was negotiating with Britain for sovereignty over the small islands. To the Argentinians, the long negotiations did not appear useful in altering British control of the islands. Furthermore, to Britain, the negotiations reaffirmed its control primarily because the 1,800 English-speaking inhabitants "were opposed to being ruled by a Spanishspeaking country whose government was an arb.trary military dictatorship."<sup>1</sup>

Three days after the Argentine invasion, on the fifth of

April, Britain dispatched the first naval elements of a task force it would assemble to retake the Falklands. This task force stopped briefly at Ascension Island, halfway between Britain and the FallCands, to await the outcome of further negotiations to perspace Argentina to withdraw its troops. When these negotiates of failed, the stage for the land war in the Falklands was set. (See Figure 1, page 53.)

On 25 April, a party of Royal Marines and Special Forces landed on South Georgia and after a short fight received the surrender of the Argentinian garrison. This successful attack gave the British task force an additional land base, albeit still some 800 miles from the Falklands.<sup>2</sup>

The eighth of May saw 3 Commando Brigade, Royal Marines, reinforced by two attached battalions of the British Army's Parachute Regiment leave Ascension with the mission of retaking the Falklands.<sup>3</sup> (The British refer to the 2nd and 3rd battalions of the Parachute Regiment as 2 Para and 3 Para respectively.) On 12 May, the British Army's 5th Infantry Brigade sailed from Britair to join the task force in the South Atlantic. On 21 May, 3 Commando Brigade and the attached parachute battalions landed at San Carlos on East Falkland. (See Figure 2, page 54.)

Although raids were conducted on Pebble Island, located just north of West Falkland, the focus of British attention during the campaign was on East Falkland. There the overwhelming preponderance of Argentinian strength, nearly 10,000 soldiers, was located. The British assumed that the

decisive battles in the Falklands Campaign would be on East Falkland and they were proven correct.

The breakout from the San Carlos beachhead started on 27 May. In the north 45 Commando headed for Douglas Settlement and 3 Para thrusted towards Teal Iniet.<sup>4</sup> 2 Para attacked towards Darwin and Goose Green. In fierce battles at Darwin and Goose Green on 28 and 29 May, 2 Para overcame strong Argentinian defensive positions, killing some 250 Argentinians and capturing over 1600 prisoners. This defeat deeply affected the morale of the remaining Argentinians garrisoned in the Falklands.<sup>3</sup> At the same time, 45 Commando took Douglas and clashed with Argentinian troops in the Mt. Kent area. Men and weapons were airlifted on to Mt. Kent which was then captured. Mt. Kent dominated the western and northern approaches to Port Stanley, the capital of the Falklands.<sup>4</sup>

On 30 May, advance elements of 5 Infantry Brigade arrived at San Carlos. By 2 June, the complete brigade had been deployed ashore in preparation for the final advance against Port Stanley. On 14 June, after a series of engagements preliminary to a final assault on the capital, the Argentine forces collapsed. A truce was arranged and the Argentinian commander, General Menendez, agreed to surrender all Argentinian forces in the Falklands Islands. The war in the Falklands was over.

A Brief Review of the Combat Service Support of the Falklands Campaign

The challenge of providing combat service support to the British land forces in the Falklands Campaign was clearly a formidable one. The sheer distance from the United Kingdom made it so. There were no contingency plans for logistic support of combat operations in the Falklands and, at the time of the task force's departure, it was unclear what form combat operations would take.<sup>7</sup> Moreover, the British had only one intermediate staging base, Ascension Island, which is located midway between Britain and the Falklands and, therefore, lay 4,000 miles from the battlefields.

Combat service support (CSS) operations on East Falkland were the responsibility of the Commando Logistic Regiment, Royal Marines, later augmented by army CSS units. These CSS operations started with the landing of 3 Commando and continued throughout the campaign.<sup>4</sup> The Logistic Regiment began its CSS planning using a basic logistic concept of operations which had been exercised the previous year.<sup>9</sup>

Although the land force fought throughout the campaign on what Hastings and Jenkins aptly described as "shoestring resources," CSS planners and tacticians adjusted and synchronized plans to insure successful sustainment of combat operations.<sup>10,11</sup> Ali in all, the British Secretary for Defense correctly described CSS in the Falklands when he said it was "a major success" of the campaign.<sup>12</sup>

Since the focus of this research is the medical support provided in the campaign, overall CSS in the Falklands will be examined further only briefly and with respect to the eight CSS principles identified in the previous chapter.

First, with respect to the principle of Support Forward, it is clear that the British followed this principle despite great difficulties in doing so. Since this was an island campaign 8,000 miles from Britain, the British had to rely significantly on seaborne support. Due to the lack of air superiority, the time supply ships could be off-loaded was limited to only a few hours each night because the ships had to make for the open sea before daybreak.<sup>13</sup> Nevertheless, forward support was maintained on the beaches initially and further inland as the ground forces advanced. The lack of sufficient numbers of helicopters was the only serious deficiency in the forward support of the ground forces.

「古中になる」

Continuous Planning was maintained once the British government decided to commit a task force to retake the islands. Planning was not perfect but adjustments were continuously made in anticipation of the requirements of tactical contingencies. Practice Economy was essential due to the limited number of support vessels available and the long transit time. Additionally, the British made Maximum Use of Standing Operating Procedures by using the time sailing to the Falklands to review and practice the procedures later to be used in the actual CSS operations.

There are many examples of Tailored Support for the land

forces in the Falklands Campaign. For instance, it was recognized that the islands' terrain was easentially either mountainous or boggy and would, therefore, necessitate a reliance on footborne or heliborne movement. For this reason, the forces committed were tailored to leave useless vehicles behind in Britain.

The British CSS planners, as well as the tacticians, practiced Centralised Control/Decentralised Operations. Despite the oversupervision which can often be found in a campaign like this one with the availability of advanced communications technology, the chain of command in the Falklands Campaign concentrated on providing resources to the soldiers on the ground and allowed the leaders on site to execute tactical and combat service support operations.

The British also followed the last two of the basic CSS principles identified in the previous chapter. They made Maximum Use of Throughput Distribution and Maximum Use of Local Resources. Ammunition, food, and other supplies were flown as far forward as helicopter assets allowed while local fuel supplies and transportation assets were used whenever available and warranted by tactical necessities.

The British soldiers who provided the vital combat service support in the Falklands Campaign did an extraordinary job. Despite limited time to plan and limited resources with which to execute support operations, the CSS provided to the land force was vigorously rendered and did not violace any of the cardinal principles of CSS doctrine.



•

•

FIGURE 1



-.

# Medical Support in the Falklands Campaign

The remainder of this chapter concerns the actual medical support of the British Army during the Falklands Campaign. The planning, execution, command and control of the medical support of the British Army units on East Falkland is described below. The chapter ends with a review of medical lessons learned and other conclusions about the Falklands Campaign medical support.

# The Planning of the Medical Support

The medical planning for the Falklands Campaign began immediately following the Argentine invasion. Although the Royal Navy had overall control of the conduct of the campaign, the Royal Navy Medical Department quickly established liaison with the medical services of both the British Army and the Royal Air Force.<sup>14</sup> The medical planning and the organization for medical support "evolved during the first few weeks until it became a fully 'corporate' effort involving all three services."<sup>15</sup>

The time required to move land forces to the Falklands allowed adequate organization of personnel, units, and supplies. The planned medical support from the Army included the following units and personnel:

(1) The 16 Field Ambulance RAMC to provide second line medical support for the land force;

(2) Surgical teams drawn from 16 Field Ambulance (Parachute Clearing Troop) and 2 Field Hospital RAMC; and

(3) Regimental medical officers assigned to each major unit (at later stages in the campaign a second medical officer was assigned to some units).16

Additionally, the British appreciated early "the difficulties of resupply for an operation of unknown duration, in which casualty figures could only be estimated,..." and made arrangements for airlift of medical supplies to Ascension Island and sealift to the Falklands.<sup>17</sup> Medical planner: also used the time to refine their planning as operational plans changed. They covered as many contingencies as possible and yet realized that adjusting to changes, or "hot planning" as they called it, would inevitably be required.<sup>18</sup>

The nearly three-week voyage to the Falklands was also put to good use in medical training and physical fitness training. Refresher training for RAMC personnel was conducted and extensive first-aid training for infantry soldiers was provided.<sup>19</sup> Medical training for infantrymen stressed immediate resuscitation, essential treatment for shock, control of hemorrhage, application of first-aid dressings, and the administration of morphine. Physical fitness received great emphasis and vigorous physical training continued aboard ship in accordance with the high

standards of the British Army as well as the rigorous requirements established for the soldiers of the Parachute Regiment.<sup>20</sup>

The British also recognized that the fighting in the Falklands would be different from that in Northern Ireland where medical treatment of the highest order was readily available. The British stressed buddy-aid in combat but also emphasized that mission accomplishment - not casualty treatment - was the first duty of an infantryman. Treatment and evacuation resources were expected to be scarce especially compared to the immediate treatment capability available in Northern Ireland.

According to British medical operational concepts, highly trained field medics were to accompany each combat unit to provide treatment beyond the average soldier's capability.<sup>21</sup> Regimental aid posts were to be positioned just behind the maneuvering infantry units. Medical planners realized that the poor terrain and the wet weather meant that casualty evacuation by road was impossible so they anticipated that the modes of evacuation would be by stretcher or by helicopter. Also, field surgical teams (consisting of a surgeon, anaesthetist, resuscitation officer, four operating theatre technicians, a blood transfusion technician and a clerk) were to be placed as far forward as possible, moving successively forward with the casualty collecting sections from the field ambulance as the combat units and the aid posts advanced.

16 Field Ambulance was to establish a dressing station and casualty collecting sections ashore. Augmented by the field surgical teams, it was to serve as an intermediate link between the surgical teams and the hospital ship in the initial phase of the campaign. Due to the lack of air superiority, operational-level planners did not want 2 Field Hospital ashore but retained aboard ship with the plan for casualties to be evacuated by helicopter from the dressing and collecting stations to the hospital afloat (a cruise ship, the SS Uganda, which had been converted into a hospital ship).22 Plans were made for 2 Field Hospital to deploy should the operational situation permit it.<sup>23</sup> Some surgical elements of 2 Field Hospital were deployed ashore with 16 Field Ambulance to bolster the field ambulance's efforts. The British even planned a holding element to look after the anticipated large number of enemy prisoners of war.24

In addition to these preparations, the British emphasized the importance of triage as well as the use of whole blood. Sorting casualties into priorities for specialty care as soon as possible was their goal. The British aim was to "provide definitive forward surgery and resuscitation within six hours...."<sup>25</sup> To insure an adequate supply of blood, "donations" taken from the soldiers during the voyage to the Falklands were augmented by blood supplied by the Army Blood Supply Depot in Britain. "Altogether, from all sources, a total of 3262 units were provided."<sup>26</sup>

などのなか。

This discussion demonstrates that the planning of the

medical support of the Falklands clearly followed the six principles of medical support identified in Chapter Three. From the outset the British medical planners sought to insure that the medical support plans conformed to the operational and tactical plans. Similarly, they planned medical support in the closest possible proximity to combat operations using significant medical resources at every echelon to insure rapid resuscitation and treatment. The flexibility which the planners displayed has been shown in their ability to quickly adjust and "hot plan" as the campaign evolved. The mobility of the medical units was expected to be severely hampered due to the nature of the terrain, but the medical planners hoped to mitigate this by situating their medical units well forward with the maneuver units. Continuity of care was also emphasized from the beginning of planning but the British also recognized their complete reliance on stretchers and helicopters for casualty evacuation (casevac). Finally, in their medical support doctrine, they recognized the importance of control of medical units to insure effective, efficient support. Their primary weakness in the area of control in the planning of the medical support of the campaign, however, appears to be their lack of centralized control over helicopter evacuation assets. Although the medical commanders realized the importance of helicopter evacuation, the paucity of helicopters in the British Army precluded their having dedicated helicopters with trained medical crews for casevac support. 27

## The Execution of the Medical Support

As the infantry went ashore at San Carlos on 21 May, organic medical personnel accompanied them. The RAMC executed the medical support in the Falklands Campaign essentially the way it was planned except that initial planning to have only an evacuation facility at Ajax Bay had to be changed.

Although the landing was unopposed by Argentinian ground forces, intensive Argentinian air strikes at the beachhead made casualty evacuation to the Uganda uncertain.<sup>23</sup> For this reason, the medical commander vigorously insisted that hospital elements be established ashore shortly after the landing - a move which the operational-level planners were reluctant to approve at first.<sup>29</sup> In spite of the fact that he risked court-martial in his confrontations with the tacticians, the medical commander's decisive action undoubtedly helped save many lives in this phase of the campaign and later.<sup>20</sup>

As the British infant ymen advanced southward toward Derwin and Goose Green, Argentinian resistance mounted. With each combat battalion went a Regimental Medical Officer (RMO), six medical assistants, a field ambulance collecting section with eight medical assistants, and regimental soldiers (bandsmen, cooks, and HQs personnel) designated as stretcher bearers.<sup>31</sup> As the ground fighting began, the coldifficulties of casualty care became apparent.

Even with the relatively light casualties, the system of treating them in the initial stages produced difficulties. The problems of casualty evacuation to regimental aid posts and beyond were immense, depending as they did upon helicopters or stretcher parties. Most engagements took place at night on remote hillsides in adverse weather conditions Many casualties, including some who had lost limbs, lay virtually untreated for up to 5 and, in some cases, 7 hours.<sup>\$3</sup>

Due to the conditions under which combat in the Falklands had to be fought, soldiers had to rely on self aid or buddy aid immediately after sustaining a wound. All soldiers carried individual first aid kits while officers and noncommissioned officers (NCOs) carried two 15 mg morphine syrettes.<sup>33</sup>

RMOs had to be highly selective about essential medical supplies and equipment because all Regimental Aid Post (RAP) equipment had to manpacked by the personnel manning the RAP.<sup>34</sup> Following the medical plan, casualties were evacuated down from the craggy hillsides by stretcher to the RAP for further treatment and then subsequently evacuated by helicopter if medical care from a higher echelon was required. (Lack of lightweight stetchers was a problem.)

The self aid and buddy aid rendered was important in saving lives plus the cold conditions promoted hemostasis even when some field dressings were poorly applied.<sup>35</sup> After initial medical treatment by a combat soldier or medic, the casualty then began the evacuation process to ultimately get him to the level of medical care needed to provide definitive treatment.

In addition to the difficulty of evacuating patients on stretchers down hillsides over rocky crags, other conditions in the Falklands increased casualties. Although the vigorous physical training conducted before and en route to the Falklands contributed to the campaign's success, no exercise schedules could have prepared the soldiers for the three weeks they spent in the harsh climate.<sup>36</sup> The cold, damp weather coupled with a lack of heat and water for personal hygiene produced a number of cold injury casualties and severely drained the fitness of all soldiers.<sup>37</sup> Some of the soldiers who participated in the campaign had also recently been involved in exercises in Germany and had already sustained "... minor degrees of non-freezing cold injury to the feet, which undoubtedly ... " predisposed them to cold injuries. <sup>38</sup> Additionally, "because of the cold climate with everything frozen at night and snow falling several times. the soldiers failed to drink adequate fluids and thus when wounded blood volume was intensified by the dehydration." "

Furthermore, British medical officers discovered another factor impacting on the initial survivability of patients. "Current teaching in the RAMC is that a tourniquet should only be used as a last resort and the reality is that this means never."<sup>40</sup> Whereas this is probably appropriate in a country with skilled medical attention readily available, the British medical officers believed that some casualties in the Falklands simply bled to death because of the accepted practice of prohibiting the use of tourniquets.<sup>41</sup>

Medical officers also discovered that the extremely cold temperatures in the Falklands negated the benefit offered by intramuscular morphine injections for battlefield casualties. The cold severely retarded drug absorption and thus prevented any measure of morphine pain relief. Later, however, after a casualty had been evacuated and had received surgery (and the patient's body had rewarmed), the morphine induced profound respiratory depression which often required massive doses of Naloxone to reverse.<sup>42</sup>

The vast majority of patients who had to be evacuated to 2 Field Hospital or to forward field surgical teams were transported by helicopter.<sup>43</sup> Available wheeled vehicles were used whenever possible also but the terrain rarely permitted this. The Army's Air Corps and the Naval and Royal Marine pilots did a fantastic job of casualty evacuation despite the difficult weather and inadequate night vision equipment.

The helicopters, since they were not solely for casevac use, usually carried ammunition in and casualties out. Of the 783 British soldiers wounded, an estimated 400 casualties were extracted by helicopters from forward positions.<sup>44</sup> Helicopters were frequent targets of both small arms fire and Argentinian aircraft. Perhaps due to the lack of medical markings, at least one helicopter engaged in a casualty evacuation mission was shot down by an Argentinian close support aircraft. Army and Royal Marine helicopter pilots performed casevac for every engagement and were decorated for their determination and skill in this role.
But there were definite problems with the aeromedevac in the Falklands. Since helicopters were also performing logistical missions, they were not always readily available for casevac. When they were available, weather and night vision difficulties as well as the lack of air superiority often made them slow in arriving. John Frost, reporting on the casevac problems of 2 Para, points out that the soldiers had difficulty getting the patients to helicopter pick-up points. They were also short of stretchers and had to improvise with ponchos and other field expedients.<sup>45</sup>

Because of all these difficulties, the British were often unable to achieve their goal of getting their casualties to surgical care within six hours of wounding. It was not unusual for evacuation to take eight hours to days.<sup>44</sup> Consequently, soldiers with major injuries often died before they could be evacuated to a hospital on land or at sea.

Once the casualties got to the hospital at Ajax Bay they were almost certain to be saved. Of all the wounded who reached hospital facilities alive, only three subsequently died. 2 Field Hospital performed admirably despite the following problems: (1) insufficient light for major surgery, especially abdominal; (2) insufficient heat for drying out damp items and properly treating hypothermia patients; (3) insufficient potable water for routine washing, insufficient sterile water for intravenous fluids or operating room irrigation, and lack of water for washing hospital linens;

(4) inadequate communications - no warning of new casualties
en route; first knowledge was upon arrival at the hospital;
and (5) insufficient paper and writing material to maintain
patient records.<sup>47</sup>

Having been established ashore at San Carlos earlier than planned, the Ajax Bay hospital was set up in a refrigerator plant.<sup>44</sup> Since the refrigerator plant was also being used to store ammunition and since it had not been marked with red crosses, the Argentinians bombed the plant within 24 hours of the landing. Even though two unexploded bombs were found in the plant after the Argentinian air raid, the medical unit continued to operate while the bombs were being defused. In one 48-hour period, the hospital performed 100 operations.<sup>44</sup>

The British were fortunate in being able to rely on the experience of the hospital's senic: surgeon and anesthetist who had set up field equipment and had operated in adverse circumstances before.<sup>10</sup> The Ajax Bay hospital was operational from 21 May to 9 June and treated 450 of the 783 casualties in the campaign. After casualties had received treatment at the hospital, they were evacuated to the Uganda if required. As mentioned above, the lighting for surgery at the Ajax Bay hospital was very poor. Consequently, the British found that about half of the abdominal wounds explored there had to be reexplored aboard the hospital ship because holes in bowels had been missed or other problems had been overlooked.<sup>51</sup>

Insuring adequate rest for the hospital personnel was a problem since everyone came on duty when patients arrived, rather than those off-shift continuing their rest. It was also noted that hospital personnel needed to be the highest caliber. It was found that a written, understood policy was a must. The examination of patients required the removal of all clothes, otherwise small wounds of entry into the back or into a thigh with significant abdominal damage could be missed because the wound of entry was never noted. The British found it very practical and appropriate to also designate individuals as resuscitation officers and triage officers and to use these same individuals in those positions throughout.<sup>32</sup>

When the intermediate objectives of the campaign were seized (Darwin and Goose Green), forces shifted eastward toward Port Stanley. Not only was Stanley the capital of the Falkland Islands, it was also where the bulk of the Argentinian forces were located. Similarly, while the second half of the campaign prepared to begin, medical resources also displaced to keep up with the advancing infantry. Like the British infantry, who "yomped" or foot marched with packs weighing up to 140 pounds, the RAP personnel yomped their equipment and supplies forward. Field surgical teams also moved forward to support the eastward engagements and to do so they established themselves at Fitzroy and Teal Inlet.

As other infantrymen sailed in the Sir Galahad to Bluff Cove to be positioned for the assault on Stanley and the surrounding Argentinian positions, elements of 16 Field Ambulance accompanied them. As the troops were disembarking, the weather cleared unexpectedly and the ship was attacked by Argentinian aircraft. Among the fifty soldiers killed

in the raid were three members of the 16th, including the second in command.<sup>53</sup>

The Sir Galahad bombing produced 179 casualties alone.34 This incident, the only one in which the available medical facilities were almost overwhelmed, was the closest the British came to a mass casualty situation. Triage had been applied at every point in the casualty evacuation chain throughout the campaign and the dental officers who filled the role as triage team leaders did a fine job. Now, with this enormous number of casualties, the experienced triage officers faced their most difficult test. Casualties from the bombing were taken to the field hospital if they required surgery or to the Uganda if they were suffering from exposure. 53 Due to the large numbers of untreated casualties which arrived at the advanced dressing station, patient transfer to the Uganda was expedited and every medic available worked continuously until all patients had been treated, released or evacuated. \*\*

The final assaults on the Argentinian positions surrounding Stanley were fierce, small unit engagements. Although casualties occured from gunshot wounds and artillery fire, treatment facilities were in place and evacuation, though still constrained by limited helicopter assets, proceeded relatively smoothly. Hospital personnel, having become proficient at battle surgery, focused on the definitive care needed to fight possible infections in the gunshot wounds and burns of the British casualties.

Once patients were stabilized aboard the Uganda, they were transferred to one of three ambulance ships. The three ambulance ships - the Hecla, the Herald, and the Hydra carried the casualties to Montevideo in Uruguay where they were transferred to Royal Air Force aircraft for the final leg of their return trip to the United Kingdom.

#### Command and Control of the Medical Support

During the Falklands Campaign effective command and control of the medical support was maintained with the exceptions of control over and communications with the casevac helicopters. The medical officers in command showed that the British medical support doctrine was sound and that their medical support system could support tactical operations thoroughly and vigorously. The medical commanders knew their personnel, equipment, procedures, and communications.

The British task force consisted of elements from all three services and the tri-service med<sup>4</sup>cal cooperation, like the tactical cooperation, was excellent. Initial planning was the responsibility of the Royal Navy, but the Royal Navy Medical Department understood how essential it was for the medical officers of all three services to actively control the vital parts they each must play in the Falklands

Campaign. Furthermore, each medical service realized the necessity of fully understanding the responsibilities and capabilities of the other two.<sup>57</sup>

With respect to command and control, the British stressed in their after action report the importance of the medical commander on the ground being able to deploy his resources to best advantage to provide the clinical links between forward medical support units and rear-area hospitals.<sup>34</sup> This was clearly demonstrated at the beginning of the land campaign when the first soldier landed and the medical commander insisted upon establishing surgical capability ashore.

The only instance of a lack of command or adequate control in the medical support arena concerned helicopter evacuation. The RAMC commanders recognize that rapid evacuation from point of wounding to surgery "is the most important factor in the saving of lives."<sup>59</sup> Yet their lack of dedicated casevac helicopters resulted in no direct control in this area. Procedures for requesting aeromedevac were cumbersome in that they had to be forwarded all the way to Commander Amphibious Warfare Headquarters for approval.<sup>60</sup> The lack of control over casevac helicopters and the scarcity of helicopters meant there were insufficient helicopters to evacuate all casualties and lives were lost because of the delay.<sup>41</sup>

69

#### Summary and Lessons Learned

The medical support of the Falklands Campaign in many ways serves as a textbook example for medical support of a rapidly deployed land force. The cooperation between the medical departments of all three British military services also serves as a fine example of joint military support operations. The operational concepts of triage, support in echelons, combat medicine, evacuation, and command and control espoused by the Royal Army Medical Corps were effectively pl\_nned and efficiently implemented. British medical support was provided in accordance with the RAMCindorsed operational concepts and it resulted in more than 99 percent of those receiving injuries which were not immediately fatal surviving to return home.<sup>42</sup>

The RAMC also closely followed the six principles of medical support: (1) conformity; (2) proximity; (3) flexibility; (4) mobility; (5) continuity; and (6) control. These principles, integral to British medical support doctrine, were used in the planning and execution of the Falklands Campaign medical support. They were applied in the context of the RAMC operational concepts already reviewed. And, they were applied effectively in all respects except for control and in only one important aspect - the control of evacuation assets.

The British field medical support system includes organic evacuation assets at every echelon of medical support

and in the Falklands it even included a hospital ship with three supporting ambulance ships. But, by doctrine and in actual practice in the Falklands, British medical commanders did not control helicopter evacuation assets. The lack of control over the evacuation helicopters was identified by the British as the main lesson for the RAMC in the Falklands. They concluded that the medical services must have dedicated helicopters under their control at all times to transport the wounded.<sup>63</sup>

There were a number of other important lessons learned about medical support in the Falklands. These included the importance of physical fitness; the value of self aid, buddy aid, tourniquets and far forward resuscitation; and the need for greater emphasis on fluid intake by soldiers. The value of simple surgical procedures and simple clinical policies which are understood by all was reaffirmed. 4 Additionally, the British Army surgeons recognized the need for broader training for physicians which would include greater familiarity with field medical equipment and battlefield casualty management. 53 The problems at the hospital at Ajax Bay (insufficent lighting, heating, and supply levels) were also noted. Further, it was reaffirmed that injuries of the head and trunk (20 percent of the casualties) are the most taxing surgical problems and that, to handle these and the other injuries in war, the military surgeon must be a truly general surgeon.66

A final point to consider, and perhaps the most

important one of all, is the element of time. The Falklands Campaign was conducted rapidly and the medical support, like the rest of the combat service support, had to be quickly but carefully assembled. With no existing plan for a contingency operation in the Falklands, the British used the limited time they had to prepare a plan that proved much more than just adequate to meet the medical support needs of the campaign. They used all the time available to increase their chances for success and to insure the actions of one service complemented those of the other two.

The Falklands Campaign, truly a joint operation, was conducted with all three medical departments working closely together. It clearly illustrates the significance which the British place on cooperation and the thoroughly professional manner in which they plan and execute medical support of combat operations.

#### ENDNOTES

#### CHAPTER IV

1. John Spanier, <u>Games Nations Play</u> (New York: Holt, Rinehart and Winston, 1984), p. 60.

2. Interview with Colonel R.J.B. Heard, British Liaison Officer, U.S. Army Command and General Staff College, Ft. Leavenworth, Kansas, 24 March 1987.

3. Max Hastings and Simon Jenkins, <u>The Battle for the Falklands</u> (New York: Norton, 1983), p. 176.

4. Interview, COL Heard, 24 March 1987.

5. Ibid.

6. Ibid.

7. Valerie Adams, "Logistic Support for the Falklands Campaign," <u>Journal of the Royal United Services Institute</u> for <u>Defence Studies</u> 129 (September 1984): 44.

MAJ Kenneth Privratsky, <u>British Combat Service Support on</u>
 <u>East Falkland: Considerations for Sustaining Tactical</u>

<u>Operations in Remote Areas</u>. School of Advanced Military Studies, U.S. Army Command and General Staff College. (Fort Leavenworth, Kansas: 25 November 1985): 3.

9. Adams, "Logistic Support for the Falklands," p. 47.

Hastings and Jenkins, <u>The Battle for the Falklands</u>, pp. 319-320.

Privratsky, <u>British Combat Service Support on East</u>
 <u>Falkland</u>, p. 16.

12. "The Falklands Campaign: The Lessons," Presentation by Secretary of State for Defence to Parliament (London: Her Majesty's Stationery Office, December 1982), p. 25.

13. COL F rry C. Summers, "Yomping to Port Stanley," <u>Military</u> <u>Review</u> 64 (March 1984): 5.

14. Surgeon Vice-Admiral Sir John Harrison, "Naval Medicine in the Falklands Conflict, April-July 1982, Overall Po'icy and Operations," Transcript of the Medical Society of London 84 (November 28, 1983): 76.

15. Surgeon Commander A.R. Marsh, "A Short But Distant War the Falklands Campaign," Journal of the Royal Society of Medicine 76 (Covember 1983): 972.

74

4 F. J. B. S. B. S.

12 15 17 17

4 4 4 4 4 4

16. D.S. Jackson, C.G. Batty, J.M. Ryan, and W.S.P. McGregor, "The Falklands War: Army Field Surgical Experience." <u>Annals</u> of the Royal College of Surgeons of England 65 (September 1983): 281.

17. Marsh, "A Short But Distant War," p. 972.

18. "Falkland Islands Campaign Medical Aspects," Briefing Script prepared by the Royal Army Medical Corps presented to Dr. Robert H. Mosebar, Medical Officer, Directorate for Combat Development, Academy of Health Sciences, Fort Sam Houston, Texas: 3.

19. Elizabeth J. Sherman, "In Bitter Little Falklands War, Enemy Medics Coopersted," <u>Army Times</u> 20 (December 29, 1986): 11.

20. Major Jonathan Bailey, Royal Artillery, "Training for War:The Falklands 1982," 63 <u>Military Review</u> (September 1983): 59.

21. Sherman, "Bitter Little Falklands War," p. 11.

22. "Saving Lives in the South Atlantic," <u>Nursing Times</u> 78. (June 23, 1982): 1044.

23. "Falkland Island: Campaign Medical Aspects," Briefing Script, p. 5.

24. Ibid.

25. Surgeon Vice-Admiral Sir John Harrison, "Naval Medicine in the Falklands," p.77.

26. Marsh, p. 974.

27. Interview, COL Heard, 24 March 1987.

28. Marsh, p. 977.

29. Privratsky, p. 23.

30. Robert Fox, <u>Evewitness</u> Falklands (London: Methuen, 1982): 126.

31. "Falkland Islands Campaign Medical Aspects," Briefing Script, p. 6.

32. Bailey, "Training for War," p. 63.

33. "Falkland Islands Campaign Medical Aspects," Briefing Script, p. 6. 34. Ibid.

35. Marsh, p. 938.

36. Bailey, p. 62.

37. Ibid.

38. "Falkland Islands Campaign Medical Aspects," Briefing Script, p. 20.

. . . . . . . . .

• • • •

39. Dr. Robert H. Mosebar, "Lessons Learned in Lebanon and the Falklands," Briefing Script prepared by the author, the Medical Officer, Directorate of Combat Development, Academy of Health Sciences, Fort Sam Houston, Texas: 12.

40. D.S. Jackson, M.D. Jowitt, and R.J. Knight, "First and Second Line Treatment in the Falklands Campaign: A Retrospective View," <u>Journal of the Royal Army Medical Corps</u> 130 (June 1984): 80.

41. Ibid., p. 81.

42. Jackson, et al., "First and Second Line Treatment," p. 81.

43. Jackson, Batty, Ryan, and McGregor, "The Falklands War,"

p. 283.

44. "Army Air Corps," information sheet published by the Royal Army dealing with aspects of the Falklands Campaign in the files of the British Liaison Officer, U.S. Army Command and General Staff College.

45. John Frost, <u>2 PARA Falklands: The Battalion at War</u> (London: Buchan and Enright, Publishers, Limited, 1983): 66.

46. LTC John W. Harmon and COL Craig Lewellyn, Report on the Symposium on the Falkland Islands Campaign - Medical Lessons at the Royal College of Surgeons, London, England, 17-18 February 1983.

47. Dr. Robert H. Mosebar, "Field Hospital in the Falkland Islands," Memorandum prepared by the Medical Officer, Directorate of Combat Development, Academy of Health Sciences, Fort Sam Houston, Texas: 2.

48. "Lesson of Falklands: Prepare for Surprises," <u>U.S.</u> <u>Medicine</u> (February 1, 1983): 3.

49. Ibid.

50. Marsh, p. 977.

51. Mosebar, "Lessons Learned in Lebanon and the Falklands," p. 13.

52. Ibid.

·····

53. Marsh, p. 977.

54. Ibid., p. 982.

55. "Lesson of Falkland," U.S. Medicine, p. 3.

56. Marsh, p. 978.

57. "Falkland Islands Campaign Medical Aspects," p. 14.

58. Ibid.

59. Ibid.

60. Privratsky, p. 30.

61. Marsh, p. 982.

82. "Falkland Islands Campaign Medical Aspects," p. 7.

63. Ibid., p. 15.

64. Mosebar, "Lessons Learned in Lebanon and the Falklands," pp. 15-16.

65. "Falkland Islands Campaign Medical Aspects," p. 16.

66. Ibid., p. 18.

-

#### CHAPTER V

### MEDICAL SUPPORT IN THE GRENADA EXPEDITION

#### The Grenada Expedition: Background

On 13 October 1983, violence erupted on the small island nation of Grenada. The charismatic Maurice Bishop, who had seized control of the island four years earlier, was deposed by hard-liners in his own government. Bishop and those who overthrew him were Marxists who had steadily been increasing Grenada's ties with Cuba and the Soviet Union.<sup>1</sup>

The other nations in the eastern Caribbean and the United States had long been uneasy about the military and political developments on the island.<sup>2</sup> The presence of advisers from the Soviet Union, Cuba, and other communist regimes, and the invitation of Cubans to construct a large airport raised many concerns.

Observers noted that, as well as the promise of

increased tourism for the island, the new airport held the potential for being a base for long-range military aircraft. Furthermore, in view of the Grenada government's close association with communist regimes, the United States and the eastern Caribbean nations worried about a communist threat to regional stability.

Additionally, the presence of six hundred American citizens attending the St. George's University School of Medicine in Grenada increased concern that another American hostage crisis such as in Iran might occur. Consequently, on 17 October, President Reagan ordered the Joint Chiefs to begin noncombutant evacuation planning.<sup>3</sup>

After Prime Minister Bishop and three of his cabinet members were executed on October 19, the resulting breakdown of law and order, the imposed shoot-on-sight curfew, and the unpredictable power struggle placed the safety of Americans in Grenada in great jeopardy.<sup>4</sup> Due to the disorganization of the new, but highly tenuous, Grenada government, repeated diplomatic attempts to coordinate an orderly evacuation of U.3. citizens failed.<sup>4</sup> Then, on October 21, the Organization of Eastern Caribbean States (OECS), plus Jamaica and Parbados, requested that the U.S. join them in intervening in Grenada by force for the protection of the entire region.<sup>4</sup>

For these reasons, on October 21, President Reagan expanded the original mission and ordered American military forces to plan for a complete seizure of Grenada as part of a combined U.S.-Caribbean security force operation on Grenada.

Although a U.S. contingency plan for the region in which Grenada was located did exist, there was no fully developed contingency plan for occupying the entire island.<sup>7</sup> Nevertheless, Operation Urgent Fury - the codename for the American intervention in Grenada - began four days later in the early morning hours of October 25, 1983.

The 1st Ranger Battalion, 75th Infantry, airdropped onto the 10,000-foot Port Salines airfield at the southern tip of Grenada at 0536 and secured it by 0715 along with the 2nd Battalion.\* The Rangers then secured the True Blue Campus of the St. George's University School of Medicine and rescued about 130 American students.\* About the same time as the Rangers' assault, 400 Marines aboard troop helicopters attacked the Pearls airport on the island's east coast.<sup>10</sup> Later, elements of the 2nd Brigade, 82nd Airborne Division, landed at Point Salines.

Shortly before noon on 26 October, Army troops advanced northward to the Grand Anse Campus of the medical school while Marines moved south from the northern tip of Grenada. The Rangers and the Marines joined together and assaulted the Grand Anse Campus and, at 1600, rescued the remaining American students.<sup>11</sup>

On the evening of 27 October, paratroopers assaulted the Cuban headquarters locate<sup>^</sup> Calvigny Barracks and secured that objective.<sup>13</sup> Except fc neutralizing small pockets of resistance later on the 27th and early on the 28th, the combat operations on Grenada were over. Within four days,

land forces consisting of two Army Ranger battalions, some Special Forces, two 82nd Airborne Division brigades, and a Marine Amphibious Unit had secured all significant military objectives and successfully rescued all U.S. citizens.<sup>13</sup>

## A Brief Review of the Combat Service Support of the Grenada Expedition

Providing combat service support to the American land forces in Grenada was a challenge primarily because of the compressed timeframe in which the Grenada Expedition occurred. Although the Joint Chiefs had discussed logistical requirements for the evacuation operation as early as 20 October, the quickly changing mission for the land forces (evacuation of noncombatants vs. seizure of the entire island) and the strict requirements for operational security severely restricted time for CSS preparation.<sup>14</sup>

The distance of Grenada from the continental United States is over 1300 miles and this posed a significant problem also. The possession of a secure base on Barbados, only 160 miles away, and the ready availability of cargo aircraft to support the expedition, however, mitigated some of the difficulties in conducting CSS operations over such a distance. (See Figure 3, page 88.)

Combat service support on Grenada was provided by the

division support command (DISCOM) of the 82nd Airborne Division.<sup>13</sup> The CSS provided began with the landing of the paratroopers on the first day of the expedition and continued after combat operations had ceased. The 82nd DISCOM advance elements deployed in phases based on established contingency plans for rapid deployments.<sup>16</sup>

The sheer rapidity with which the expedition began and proceeded made the provision of combat service support difficult. CSS soldiers reacted quickly, like their combat arms counterparts, and adjusted plans to sustain the combat operations which ended just 96 hours after they had begun.

Since the subject of this research is the medical support provided to the Army expeditionary forces, overall combat service support in Grenada will be discussed further only briefly. Combat service support will be examined with respect to the eight CSS principles identified in Chapter Three.

The 82nd DISCOM and the organic CSS elements of the Ranger and airborne battalions certainly adhered to the principle of Support Forward. The Americans established combat service support on the island as the Port Salines airfield was being secured. They took advantage of the support which the naval forces stationed just off the island could provide and enjoyed the air superiority which the Navy maintained. (See Figure 4, page 89.)

Continuous Planning simply was not conducted with respect to combat service support, however. Concerns about

secrecy severely restricted the number of personnel included in the planning process and for this reason there were no CSS representatives for planning.<sup>17</sup> The Americans did *Practice Economy* both in their use of light forces and because of their dependence on aerial movement and aerial resupply. *Maximum Use of Standing Operating Procedures* was essential also because units, especially CSS units, had little time to prepare for the specific requirements for the Grenada Expedition.

There are a number of examples of Tailored Support for the Grenada Expedition. Since, as mentioned above. the mission called for light infantry forces, support planning was tailored to provide for the CSS needs of those forces. The restriction on the information available, however, prevented a great deal of support tailoring and forced reliance on standard, general support packages.

Although American CSS commanders usually rely on Centralized Control/Decentralized Operations, Grenada posed a real problem in this area. This was due again to the operational planners not including the CSS planners in the planning process. Inaccurate intelligence and lack of understanding about the capabilities of different services in this joint operation degraded the amount of control CSS commanders had in employing their support packages. This problem was not confined to CSS operations only. Indeed, the command structure of the Grenada Expedition was judged to have hampered every facet of the operation.<sup>19</sup>

Due to the short duration of the combat operations in Grenada, Maximum Use of Throughput Distribution was only tested to a limited degree. Even though American CSS officers could rely on Barbados and Roosevelt Roads, Puerto Rico as intermediate support bases, the relative proximity by air of Fort Bragg, North Carolina (1750 miles) did facilitate throughput distribution.<sup>19</sup> The 82nd DISCOM and the 1st Corps Support Command, both based at Fort Bragg, were linked by existing contingency plans and, consequently, were wellsuited to provide throughput CSS to the U.S. forces on Grenada.<sup>20</sup>

The Americans made the Maximum Use of Local Resources as was feasible. Rangers used trucks and bulldozers parked by the Cubans on the Point Salines runway to remove enemy barricades and local facilities were used for CSS operations. The rapidity of the combat operations severely restricted any significant demand on local resources, however.

The American soldiers from the 82nd DISCOM and the 1st Support Command used established combat service support procedures to successfully support the Grenada Expedition. In view of the rapidly changing situation, the CSS soldiers displayed great initiative. Even with significant problems in the operational command structure, they reacted quickly and adjusted effectively. The violations of the principles of *Continuous Planning* and *Centralized Control* were caused by the significant deficiencies in the joint command structure rather than any CSS procedural or organizational problems.



•

•

FIGURE 3



.

.

172

127

# GRENADA



#### Medical Support in the Grenada Expedition

The rest of this chapter deals with the medical support of the American Army during the Grenada Expedition. The planning, execution, command and control of the medical support of the American Army units in Grenada is described below. The chapter ends with a review of medical lessons learned and other conclusions about the Grenada Expedition medical support.

#### The Planning of the Medical Support

As mentioned earlier, combat service support planning was essentially not conducted for the Grenada Expedition. Indeed, "critical details of the support plan, combat support and combat service support were not available during the planning phase."<sup>21</sup> Consequently, medical support planning had to be conducted with little information. The restrictions of operational security imposed at every echelon of command prevented medical planners (both the 82nd Airborne Division surgeon and the 307th Medical Battalion commander) from being informed of the expedition's destination until after combat operations had begun.<sup>22</sup>

Although the Commander-in-Chief Atlantic (CINCLANT), Admiral McDonald, had overall command of the operation, there was never an overall medical command or control element

established for the operation by the Navy or any other service. The quick formation of Joint Task Force (JTF) 120 (commanded by Vice Admiral Metcalf) and the wide range of land forces it involved (Marines, Rangers, paratroopers, and Special Forces) is still an object of controversy.<sup>23</sup> The rapidly formed coalition of forces was not synchronized and the medical support, like most aspects of the expedition, lacked centralized control. ۲.

.

Since the Army forces committed to the Grenada Expedition were moved by air, time was scarce to specifically organize medical personnel, units, and supplies. Standard rapid deployment packages had to suffice.<sup>24</sup> The units normally designated for supporting the Army forces were alerted and included the following:

(1) The 307th Medical Battalion, 82nd Airborne Division, to provide division level medical support for the Army elements; and

(2) Medical platoons organic to the Ranger and airborne battalions.

In addition, the Commander of the 82nd Airborne Division was told by CINCLANT that the U.S.S. *Guam* and the U.S.S. *Saipan* had significant medical support capabilities (equivalent to a 100-bed or larger hospital). For that reason, the division commander believed that medical support units from the division could be kept to a minimum.<sup>25</sup>

The pre-planned medical support packages which the medical planners would use were well-suited to the medical support mission. They consisted of two echelons from each company in the 307th Medical Battalion. These two echelons were designated Alpha and Bravo echelons respectively. The Alpha echelon was a light, air-droppable package designed to go in with the initial assault. The Bravo echelon was a heavy package to be airlanded with follow-on elements.<sup>26</sup>

There was no time available for special medical training even if the medics had been fully briefed on what to expect. The medics were already at a high state of readiness, however. Many of them had completed emergency medical technician training through on-going medical proficiency training programs.<sup>27</sup> The physicians assigned for the operation were fully qualified Medical Corps officers but none had attended the AMEDD's Combat Casualty Care Course, called C4.<sup>28,29</sup> C4 is designed to prepare AMEDD officers to function successfully at forward points in the battlefield casualty care system.<sup>30</sup>

In accordance with American medical operational concepts, a physician, a physician assistant, and medics were to accompany each Ranger and airborne battalion committed to combat. Battalion aid stations were to be established as close as possible to where infantrymen were in contact with opposing forces.

South Second

Since information on what to expect with regard to the intensity of the conflict was so sketchy, medical planners

once again had to rely on existing, general medical support plans. These called for a number of ground and air ambulances to be deployed as soon as aircraft to transport them to Grenada were available. The ground ambulances were organic to the airborne battalions and the medical companies of the 307th Medical Battalion. The air ambulances were to be provided by the 57th Medical Detachment (Air Ambulance), a corps level medical unit.

The medical companies in the 307th Medical Battalion had a small, organic field surgical capability in the Alpha echelon so priority for movement was planned for the men and equipment providing that capability. If required to support the American combat operations, additional medical company personnel and equipment in the rest of the Alpha echelon and the Bravo echelon would be flown to Grenada.

Division level medical care for the Rangers' initial assault would not be immediately available from the Army. That level of care would be provided by the Navy until Company C of the 307th arrived with the 82nd Airborne Division soldiers who were to be airlanded at the Port Salines airfield. Ostensibly the Navy could provide division level care on an interim basis. However, "no Army reference could be located that provided for planning data for Navy shipboard medical facilities."<sup>31</sup>

In preparing for the Grenada Expedition, medical planners attempted to follow the six principles of medical support. They were frustrated in their attempt, however,

because of the secrecy surrounding the operation. Unable to find out specific information about operational and tactical plans, they had to assume their standard medical support packages would conform to the medical needs of the combat units employed. They also had to assume that medical support elements would be allowed to deploy with the combat elements they supported thus permitting close proximity to the soldiers in contact with opposing forces.

Lack of operational information prevented flexibility in planning yet flexibility was intrinsic to the standard rapid deployment packages. The mobility of the medical support units was insured by the configuration of the medical support packages also. But the medical planners could only assume they would get the deployment airlift they would need at the time they needed it.

Maintaining continuity of care was covered in existing medical contingency support plans but the 82nd medical planners did not know when the Rangers would go in or how soon afterwards the 82nd would land. Consequently, continuity of care in that instance would depend on the Navy. Also, how great the need for medical care would be and how the system for evacuation of casualties would work during the combat operations could not be forecasted by the planners.

Finally, control of the medical units to be used was relatively clear for each service, but the interrelationships between medical elements of each service were not. The lack of a unified medical plan prevented medical participants in

the Grenada Expedition from knowing what medical elements were to be involved and what capabilities each had. In short, if synchronization of medical support would occur, it would happen only if existing contingency plans and standing operating procedures made it happen. Joint planning of medical support and casualty evacuation did not occur. And the Army medical planners did not have the time or the access to the information to resolve this serious deficiency.

#### The Execution of the Medical Support

As mentioned earlier, the Rangers began the attack on Grenada at 0536 on 25 October 1983. Due to an "umbrella of flak" over the Port Salines airfield, the C-130 aircraft carrying the Rangers were forced to dive low and the Rangers had to jump from only 500 feat, "a height not employed in combat since World War II."<sup>3</sup> (Oddly enough, despite the strict operational security maintained with the Army, the Cuban government was given two hours advanced warning of the attack by the State Department. This was apparently done to give the Cubans on Grenada time to withdraw if they wished, but instead Premier Castro exhorted his soldiers to fight to their deaths.<sup>33</sup>)

Shortly after the initial assault on the airfield, the Rangers' organic medical personnel parachuted in and provided immediate treatment to a small number of casualties. The Ranger medics received some medical support from the Navy but essentially treated the casualties themselves until the advanced element of Company C, 307th Medical Battalion (consisting of one orthopedic surgeon, one enlisted practical nurse, and four medical aidmen) arrived approximately twelve hours later.<sup>34</sup> Once the Rangers had secured the airfield, paratroopers from the 82nd airlanded.

As the first airborne battalions from the 82nd arrived at the Port Salines airfield and the division commander, Major General Trobaugh, judged the Cubans to be giving much greater resistance than expected, "the decision was made to send more combat and combat support forces to Grenada instead of continuing the deployment of the combat service support elements."<sup>33</sup> The commanding general's decision was undoubtedly influenced by his belief that the Navy could provide any medical care needed.

'Company C's advanced element arrived as early as it did because the surgeon and his medics deployed with one of the lead airborne battalions. Due to severe limitations on aircraft space availability, the surgeon had only been able to bring basic life saving equipment for stabilizing patients. He and his five personnel had to use "shelters of opportunity" to establish their limited field surgical section on the evening of the 25th.<sup>36</sup>

96

たちろうどうちに、これとしていた。

Although the night of the 25th saw no significant combat action, the Rangers and the paratroopers did receive sporadic small arms fire around the airfield. Except for the evening attack that would occur on the 27th, the Grenada Expedition's combat operations would involve no attacks at night. Enemy sniping continued each night until combat operations concluded, however.

By the morning of the 26th, it was clear to the Army medical personnel at the division level of care that the Navy's actual medical capabilities did not match the information transmitted to the 82nd Division headquarters. Indeed, neither of the ships identified as being able to provide medical support had a full medical staff and the U.S.S. Guam was the only one of the two ships with a surgeon. Fortunately, another element of Company C's Alpha echelon was expected to arrive soon.

The Alpha echelon had been broken down into three parts because of the changes in chalks, or loads, for the Grenadabound aircraft. The first part had arrived as already described. To insure the deployment from Fort Bragg of the remaining two parts of the Alpha echelon was not postponed, the 307th Medical Battalion commander had to personally intervene.<sup>37</sup> Unfortunately, the second part, consisting of two additional physicians and a nurse anesthetist, was diverted from Port Salines due to small arms fire around the airfield. Those personnel and their equipment had to land on Barbados and were unable to get to Grenada until the evening

of the 27th. The third part, with another physician and fourteen more medical personnel, ended up preceding them and arrived around 1200 on the 26th even though they had to circle the airfield for several hours before landing.

Shortly before that part of the Alpha echelon arrived on the 26th, the Rangers and paratroopers began their northward advance to the Grand Anse Campus of the St. George's University School of Medicine. Small numbers of casualties were received at the Company C clearing station that afternoon from that action. Since few ambulances had been deployed early, other tactical vehicles had to be used to move casualties from aid stations to the clearing station. The Company C ambulances had not yet arrived either, so the medical platoons had to evacuate their casualties to the medical company instead of being able to rely on the medical company to evacuate their patients from them.<sup>33</sup>

In addition to personnel, the first part of Company C's Alpha echelon was now augmented with additional equipment, supplies, and tentage which arrived with the third part of the Alpha echelon. Although the first part of Company C's Alpha echelon was initially co-located with the lead elements of the Forward Area Support Team (FAST) headquarters, the two elements of the medical company now relocated to the other side of the airfield to allow for room to set up more equipment and lay out landing areas for air ambulances.

As the set up progressed, the Company C medics saw that the disruption of the Alpha echelon and the diversion of the

second part of the echelon to Barbados had adversely affected the medical company's ability to establish itself. Company C's Alpha and Bravo echelons were configured using a standard 307th Medical Battalion cross-loading concept.<sup>39</sup> Crossloading prescribed that no entire functional area (such as a ward, a treatment section, or the admissions and dispositions element) be loaded on any one vehicle. This was to preclude the entire loss of a functional area should an aircraft go down. But due to the break-up of the Alpha echelon during deployment, the Company C medics found that they only had parts of each functional area once on Grenada. Luckily, through the exertion of initiative and the relatively light number of casualties, Company C was able to continue care despite these problems.<sup>40</sup>

Aware of their own equipment shortcomings as well as the Navy's, the Army medical personnel were growing increasingly concerned about their ability to support combat operations. The rest of the 26th and the 27th were thus "consumed in trying to determine what was actually required in the forward area and coordinating a priority for deployment of the elements prepared for deployment."<sup>41</sup>

20

After the rescue of the students at the Grand Anse Campus late in the afternoon on the 26th, Army combat operations halted while reconnaissance elements gathered additional information on enemy positions. This lull in the fighting gave the Army medical personnel at division level additional time to establish themselves.
Early on the 27th, just in time to support the Army's attacks planned to bring the Grenada Expedition's combat operations to a close, additional medical units began to arrive. The 57th Medical Detachment (Air Ambulance) landed with three helicopters and was operational by 0700.42 Prior to this time, tactical helicopters had to be used to transport casualties in the same way tactical vehicles had been used in lieu of ambulances.

As units prepared for the final assault on the Cuban headquarters at the well-fortified Calivigny Barracks, an accident occurred which was to result in the beginning of a mass casualty situation for Charlie Company. A Navy Corsair, called in to attack the Cuban position, strafed a group of U.S. paratroopers by mistake and wounded twelve of them.<sup>43</sup> The air ambulances, which had arrived early in the day, braved enemy fire to pick up and rapidly transport these casualties to Company C's clearing station at Port Salines.

Then, in what was certainly propitious timing, the other two physicians and the nurse anesthetist from Charlie's Alpha echelon arrived in the evening of the 27th after having been diverted and stranded on Barbados the day before. This was also important because additional casualties would be generated as the assault on Calivigny Barracks began around 1800.44

Although enemy forces were estimated to be as high as 400 around the Cuban headquarters, the actual number was much smaller. Rangers and paratroopers, transported in 82nd

100

\*\*\*\*\*\*\*\*\*\*\*\*

Airborne Division Blackhawk helicopters, attacked and secured the entire area by 2100.45 In the assault, however, two helicopters collided and the resulting casualties, in addition to the casualties from the fighting, threatened to overwhelm Company C's capabilities.44

In the space of two hours the Alpha echelon received thirty casualties. Using all the medical resources at hand, to include some medics from the Ranger battalions, the medical company handled the crisis. In addition to quickly treating and stabilizing a number of patients, the senior surgeon coordinated with the Air Force to fly four patients to the hospital at Roosevelt Roads, Puerto Rico. Since the Air Force's 1st Aeromedical Evacuation Squadron had not yet been ordered to deploy, he had to send a physician and nurse with the patients, however.<sup>47</sup>

Shortly thereafter coordination was made to have the Air Force evacuate casualties to the Roosevelt Roads Naval Hospital in Fuerto Rico. There patients were temporarily Hospitalized and prepared for evacuation to Brooke Army Medical Center or Walter Reed Army Medical Center. Patients with severe injuries who required more immediate emergency care were flown to the *Guam* or to Barbados.<sup>4,5</sup>

Although combat operations continued through the 28th to neutralize small pockets of resistance, the number of casualties dropped significantly after the mass casualty on the evening of the 27th. During the four days of the combat, nineteen soldiers were killed and 152 were wounded.<sup>+9</sup>

# Command and Control of the Medical Support

Effective joint command and control of the medical support in the Grenada Expedition was never established. Although the American armed forces had a commonly shared manual for the conduct of medical support in joint operations, neither it nor the procedures prescribed in it were used.<sup>50</sup> The officers in the medical departments of each service knew their own medical support doctrine well. But no unified medical plan brought these elements together either for planning or executing the medical support mission.

The Grenada task force consisted of units from all three services. As the Grenada Expedition rapidly evolved and was executed, the soldiers, sailors and airmen cooperated well. But they had to overcome the lack of joint coordination which should have been provided by the joint commander and his staff.

The lack of initial planning to coordinate the medical support effort was a critical deficiency. Medical officers did not know what other services were involved in the expedition so, consequently, they had no opportunity to understand the capabilities of the other two medical departments. They relied on standing operating procedures and established medical packages to match requirements. Luckily, the relatively low level of casualties prevented the meager Army medical resources from being overwhelmed.

Problems in communication between elements of the 82nd

DISCOM's Forward Area Support Team (FAST) and the supported ground units initially hampered effective control.<sup>31</sup> Similarly, there was difficulty in establishing communication with the Army medical units on Grenada and the supporting Navy ships.<sup>32</sup> This lack of communications exacerbated the already deficient command and control structure of the expedition.

e e e la construcción de la constru

Medical officers had difficulty communicating with supported units as well as other medical units both in their own service and in other services. They could not ascertain if there was a plan dealing with the regulation of the flow of patients and they had no idea what the situation was with respect to patient evacuation within or outside of the combat zone. Furthermore, during nearly all of the time combat operations were being conducted, the Army medical officers had difficulty determining what was available to treat their patients other than those supplies and medical equipment items which they had brought to Grenada themselves.

Tactical commanders also failed to appreciate the need to have preventive medicine personnel accompany them on the deployment to Grenada as well as the need to follow basic measures to prevent disease. As a result, sanitation standards and water treatment procedures were largely ignored and could have jeopardized a longer operation.<sup>53</sup>

The crucial importance of command and control in the area of medical support is clear. Rapidly deployed land forces like those used in the combat operations of the

Grenada Expedition require decisive action to insure medical support is available when and where needed. The medical support of the Grenada Expedition was nearly a disaster because of excessive operational security, a lack of relevant information for planning and execution, and a failure of the joint command structure to exert its responsibilities for coordinating the medical support efforts.

The initiative of the medical soldiers in Grenada and at Fort Bragg saved the medical support effort for the Army. The impact of breaking the sequence of the deployment of the medical support packages could have been catastrophic had it not been for the resourcefulness and initiative of the medics. The flexibility which they demonstrated undoubtedly saved the lives of soldiers who otherwise might have died because of the lack of planning and the miscommunication at the joint command level.

Decisive action by medical commanders was critical to the overall success of Army medical support in Grenada. This is illustrated by two examples. The first was the intervention of the 307th Medical Battalion commander in defending the priority of the deployment of Company C's Alpha echelon. The second was the coordination of airlift for patients by the Alpha echelon senior surgeon with the Air Force on the evening of the 27th. Both examples point out the importance of active command and control in insuring the timely provision of effective medical support.

## Summary and Lessons Learned

The medical support of the Grenada Expedition was poorly planned and was just barely executed successfully. For these reasons, the Grenada Expedition medical support serves as a good case study for Army medical planners to review when examining problems in medically supporting rapidly deployed land forces.

The poor planning was the result of excessive operational security. Clearly, security is essential in an operation like Grenada, but there can be no excuse for operational commanders excludin, combat service support commanders from planning any operation. The effects of the poor planning were exacerbated by the fact that it occurred at the joint level and thus disrupted established military medical support systems which rely on interservice cooperation. The misinformation about Navy medical capabilities and the failure to activate a vitally needed Air Force aeromedical evacuation squadron nearly contributed to the death of American soldiers wounded in Grenada.

The Army medical officers relied on established medical support packages which were designed to be flexible enough to allow for the provision of minimal medical support. The disruption in the division level medical company's standard deployment configuration, however, almost prevented that level of medical support from arriving on time.

Tactical commanders, in the turmoil of the rapid

deployment and confused battle situation, lost sight of the need for far forward medical support and preventive medicine measures. They also forgot basics like insuring that their soldiers drank enough fluids or that the soldiers did not overload themselves. Consequently, they filled their battalion aid stations with heat stress casualties.<sup>54</sup>

The medics relearned some old lessons as well. First. they relearned the value of simple emergency measures like tourniquets which undoubtedly saved a number of lives in Grenada.55 Second, they learned that medics have to keep their equipment as light as possible to keep up with maneuvering infantry units. 56 Third, they saw the importance of prior coordination with medical support levels below and above their own. Fourth, they demonstrated the tremendous value of air ambulances, especially when under the direct control of AMEDD officers, and the remarkable performance of the UH-60 helicopter as an air ambulance platform. Finally, they learned to functionally load their deployment packages to insure functional integrity was maintained whenever possible and to emphasize to tactical commanders the importance of maintaining the sequence of medical support package deployments. 57

The Army medical officers were nearly prevented from following the six principles of medical support by their exclusion, along with the other CSS planners, from the planning of the Grenada Expedition. Conformity, proximity, flexibility, mobility, continuity, and control were difficult

to achieve in view of the lack of information available to them. Although the medics did achieve the first five to a limited degree, control was missing from a joint standpoint and severely debilitated within the context of the Army because of faulty decisions by tactical commanders.

Grenada pointed out significant but common problems in conducting joint operations. The Army medical support difficulties were simply a microcosm of these problems, especially in such an information vacuum.

Perhaps most important of all is the lesson Grenada teaches about the need for completeness in a medical system to support soldiers in combat. Effective medical support requires availability of care across the spectrum. Whether a soldier needs the simplest or the most comprehensive care, all echelons of care must be in place in the combat zone or all levels of care must at least be accessible from the combat zone. In Grenada medical care across the spectrum of comprehensiveness was not provided and the means to access care outside the combat zone was not carefully planned. The lack of sufficient medical treatment capability on Grenada and the initial lack of an Air Force aeromedical evacuation squadron, for instance, resulted in Army medics being forced to prematurely evacuate patients on poorly equipped cargo aircraft. \$9

The Grenada Expedition was an overall success but it cannot serve as a model for joint operations. Likewise, the medical support cannot serve as a model for joint medical

support. However, the lessons Grenada teaches are invaluable to future operational and medical planners. Failure to learn those lessons could cause the medical support of future rapidly deployed land forces to end in disaster.

· · · · · · · ·

## ENDNOTES

#### CHAPTER V

Ralph Kinney Bennett, "Grenada: Anatomy of a 'Go'
Decision," <u>Reader's Digest</u> (February 1984): 72.

2. Ibid.

3. Ibid., pp. 72-73.

4. <u>Grenada: A Preliminary Report</u>, Department of State and the Department of Defense, Washington, D.C. (December 16, 1983): 2.

5. Ibid.

 Lieutenant Colonel Andrew M. Perkins, Jr., "Operation Urgent Fury: An Engineer's View," <u>The Military Engineer</u> (March-April 1984): 86.

7. <u>Strategic Studies: National Security Policy</u> <u>Considerations</u>, U.S. Army Command and General Staff College, Fort Leavenworth, Kansas (1986): 120.

8. Dan Cragg, "The U.S. Army in Grenada," Army 33 (December

1983): 29.

9. Ibid.

10. Ed Magnuson, "D-Day in Grenada," <u>Time</u> 122 (November 7, 1983): 22-23.

11. Cragg, "The U.S. Army in Grenada," p. 31.

12. Ibic.

13. "JCS Replies to Criticism of Grenada Operation," <u>Army</u> 34 (August 1984): 29.

14. Bennett, "Grenada: Anatomy of a 'Go' Decision," p. 73.

15. Captain Robert M. Radin and Chief Warrant Officer (W3) Raymond A. Bell, "Combat Service Support of Urgent Fury," <u>Army Logistician</u> 16 (November-December 1984): 17.

16. Ibid., p. 16.

and the second of the second second

17. Lessons Learned Grenada: U.S. Army Lessons Learned From 1983 Operation Urgent Fury, Department of the Army, Washington, D.C. (May 1985): IV-f-2.

18. Grenada Work Group, Operation Urgent Fury Assessment,

Combined Arms Center, U.S. Army Training and Doctrine Command, Fort Leavenworth, Kansas (9 July 1985): V-9.

19. Lieutenant Colonel Michael J. Byron, "Fury from the Sea: Marines in Grenada," <u>U.S. Naval Institute Proceedings</u> 110 (May 1984): 120.

20. Radin and Bell, "CSS of Urgent Fury," pp. 16-17, 19.

21. Major James M. Dubik and Major Terrence D. Fullerton, "Soldier Overloading in Grenada," <u>Military Review</u> 67 (January 1987): 44.

22. Telephone interview with Lieutenant Colonel Edward B. Wilson, Health Care Operations, Office of the Surgeon General, United States Army, Washington, D.C., 23 December 1986. (Lieutenant Colonel Wilson commanded the 307th Medical Battalion, 82nd Airborne Division, during the Grenada Expedition.)

23. Dorothea Cypher, "Urgent Fury: The U.S. Army in Grenada," Chapter Nine in Dunn, Peter M., and Watson, Bruce W., <u>American Intervention in Grenada: The Implications of</u> <u>Operation "Urgent Fury"</u> (Boulder, Colorado: Westview Press, 1985): 101.

24. LTC Wilson, telephone interview, December 23, 1986.

25. Telephone interview with Major David L. Nolan, Health Care Operations, Office of the Surgeon General, United States Army, Washington, D.C., 22 December 1986. (Major Nolan was the 307th Medical Battalion's operations officer during the Grenada Expedition.)

26. Letter interview with Captain Stephen S. Bradley, Plans, Operations, Training and Security, U.S. Army Medical Department Activity, West Point, New York, 31 December 1986. (Captain Bradley commanded Company A, 307th Medical Battalion, supporting the 3d Brigade, 82nd Airborne Division, during the Grenada Expedition.)

27. Ibid.

28. Letter interview with Captain Vincent E. Ashley, Academy of Health Sciences, U.S. Army Health Services Command, Fort Sam Houston, Texas, 8 December 1986. (Captain Ashley commanded Company C, 307th Medical Battalion, supporting the 2d Brigade, 82nd Airborne Division, during the Grenada Expedition.)

29. CPT Bradley, letter interview, 31 December 1986.

30. U.S. Department of the Army Regulation 351-3, "Professional Training of AMEDD Personnel," 15 December 1985.

31. Grenada Work Group, <u>Operation Urgent Fury Assessment</u>, p. X-57.

• •

32. Ed Magnuson, "Now to Make It Work," <u>Time</u> 122 (November 14, 1983): 24.

33. <u>Strategic Studies: National Security Policy</u> <u>Considerations</u>, p. 130.

34. Interview with Captain Douglas S. Phelps, Combined Arms Services Staff School, Fort Leavenworth, Kansas, 4 March 1987. (Captain Phelps was the executive officer of Company C, 307th Medical Battalion, the first medical company deployed on the Grenada Expedition.)

35. Captain Douglas S. Phelps, "Synopsis of Division Level Medical Support in Grenada," undated after action summary provided by the author.

36. CPT Phelps, interview, 4 March 1987.

37. Ibid.

38. Phelps, "Synopsis of Division Level Medical Support in Grenada."

39. MAJ Nolan, telephone interview, 22 December 1987.

113

40. CPT Phelps, interview, 4 March 1987.

41. Fhelps, "Synopsis of Division Level Medical Support in Grenada."

42. Telephone interview with Major Kevin J. Swenie, Health Care Operations, Office of the Surgeon General, United States Army, Washington, D.C., 5 March 1987. (Major Swenie was the operations officer for the 57th Medical Detachment (Air Ambulance) during the Grenada Expedition.)

43. Magnuson, "Now to Make It Work," p. 24.

44. Cragg, "The U.S. Army in Grenada," p. 31.

45. Ibid.

. .

46. Telephone interview with Dr. Robert H. Mosebar, Medical Officer, Directorate of Combat Development, Academy of Health Sciences, U.S. Army Health Services Command, Fort Sam Houston, Texas, 10 March 1987.

47. Phelps, "Synopsis of Division Level Medical Support in Grenada."

48. CPT Bradley, letter interview, 31 December 1986.

49. Richard Gabriel, "Scenes from an Invasion," <u>The</u> <u>Washington Monthly</u> (February 1986): 41.

50. U.S. Department of the Army Field Manual 8-8, "Medical Support in Joint Operations," 1 June 1972.

- 5

51. Lieutenant Colonel Kenneth C. Sever, "782nd Maintenance Battalion in Grenada," <u>Ordnance</u> 2 (Winter 1984): 6.

52. CPT Phelps, interview, 4 March 1987.

53. Dr. Mosebar, telephone interview, 10 March 1987.

54. Dubik and Fullerton, "Soldier Overloading in Grenada," p. 39.

55. D: 3cbar, telephone interview, 10 March 1987.

56. Ibid.

57. MAJ Nolan, telephone interview, 22 December 1986.

58. Telephone interview with Dr. Dobert H. Mosebar, Medical Officer, Directorate of Combat Development, Academy of Health Sciences, U.S. Army Health Services Commanc, Fort Sam Houston, Texas, 9 February 1987.

### CHAPTER VI

#### COMPARATIVE ANALYSIS

The comparative analysis of the medical support of the Falklands Campaign and the Grenada Expedition will first focus on a discussion about how well the medical support operations followed the established principles. Then, the medical support operations in the two expeditions will be compared with respect to the use of operational concepts. Finally, a comparison of the lessons learned in each operation will be presented.

### Adherence to Principles

The principle of conformity was strictly observed by the Royal Army Medical Corps in the planning and execution of the medical support in the Falklands. The British medical officers were thoroughly briefed on the basic plans for the

campaign and notified of changes as the campaign evolved. Consequently, they were not only able to react and insure that medical support conformed to tactical developments, but they were able to be proactive and anticipate how to tailor medical support to fit battlefield developments.

Although fully subscribing to the principle of conformity, the American medical officers supporting the Grenada Expedition found it difficult to execute the principle because they, along with all combat service support officers, were excluded from the operational planning. Grenada Expedition medical support, therefore, conformed only to a limited degree due to the reliance on standard deployment packages and having to overcome the disruption in standard airflow caused by tactical decisionmakers.

The principle of proximity was maintained by the British in the Falklands despite the difficult terrain. Forward medical support elements man-packed their equipment to remain close to the combat units. Hospital elements were established ashore to provide vitally needed, proximallylocated, medical support. The medical commander emphasized this point to the operational planners so they would allow him to do this in the initial stage of the campaign.

Due to the disjointed manner in which medical support elements at division level were deployed, American medical officers in Grenada found it difficult to position themselves near units in contact. Their attention had to focus on simply establishing adequate medical support initially.

Luckily, due to the short distances involved in the Grenada combat operations, proximity was less important than in the Falklands. Additionally, the availability of helicopters, albeit initially not medevacs, minimized the need for closer proximity of the division level medical support.

Although the British medics were somewhat constrained due to the lack of British air superiority in the Falklands Campaign, the forward positioning of medical resources at every echelon gave them flexibility in shifting medical resources where most vitally needed. The time available to them during the deployment to the Falklands also enabled them to insure that they had ample medical resources to give them flexibility in supporting combat operations.

In Grenada, however, the scarcity of medical resources at division level and the lack of information about the capabilities of the Navy resulted in there being ensentially no flexibility in medical support. What flexibility there was came about as a result of the personal initiative of the medical officers involved and the limited flexibility already built into the standard medical deployment packages.

Mobility was carefully considered by the British for medical support in the Falklands as well as for tactical operations. Understanding their mission and the Falklands terrain, the British prepared their units to deal with the peculiar features of that area of operations. They realized going into the campaign that they must rely either on heliborne or footborne transport.

Due to the limited time of the combat operations of the Grenada Expedition, mobility of the medical units once in Grenada was not significant. But the mobility, or lack thereof, in getting to Grenada was a severe problem. Additionally, lack of ground ambulances, both for unit and division level medical units, hampered efficient medical support. The relatively late arrival of air ambulances (two days after combat operations began) also degraded the medical units' ability to move patients, medical personnel, equipment, and supplies when needed.

Continuity of care, on the whole, was maintained by the British in the Falklands but the use of non-medical helicopters actually resulted in an interruption in continuity. British casevac helicopters, unlike American medevac helicopters, do not carry highly-trained medical aidmen and are not piloted by medically-trained evacuation aviators. For these reasons, patient care in aeromedical evacuation in the Falklands was not truly continuous. Falklands casualties, in addition to having to face the hazards of being transported in non-medically marked helicopters (remember that at least one was shot down), also ran the risk of not having anyone to deal with in-flight medical emergencies should they develop during evacuation.

Maintaining continuity, indeed, even establishing it, was a problem for American Army medical personnel in Grenada. This was due, of course, to the lack of time to plan, the disjointed deployment sequences, and the lack of any overall

medical plan for this joint medical support operation. Continuity of care was difficult to establish on Grenada and continuity during the evacuation of patients to Barbados, Puerto Rico, or elsewhere was inadequate due to limited personnel, equipment, and supplies.

As far as control, the British did a superb job achieving this principle with the highly notable exceptions of not being in control of or in communication with the casevac helicopters. Overall, their command of medical resources functioned efficiently in the joint arena and their coordination with medical elements within each service, as well as between different services, was consistently effective. Their failure to directly control aerial evacuation assets, however, remains their principal operational weakness and their primary problem in the medical support of the Falklands Campaign.

Control of medical support for the Grenada Expedition was essentially taken out of the hands of medical officers by operational planners and tactical decisionmakers. The result was a near medical disaster. The 307th Medical Battalion commander and the orthopedic surgeon who was the first division level medical officer to arrive in Grenada both had to push hard to maintain and exert the limited control left to them. The nearly disastrous consequences in Grenada point out the significance of violating this principle and the necessity for medical control of medical support operations.

# <u>Use of Operational Concepts</u>

As previously stated, the operational concepts of the USAMEDD and the RAMC encompass the following : (1) triage; (2) echelons of medical support; (3) elements of combat medicine; (4) patient evacuation; and (5) command and control.

Both the British in the Falklands and the Americans in Grenada were each faced with a mass casualty situation. Although the situations they faced severely taxed the available medical resources for a short period of time, the level of casualties never overwhelmed the medical support units, however. Consequently, while triage was used to prioritize patients for treatment, the use of the expectant category was never required. Medical resources were adequate, although just barely so in Grenada, to permit treatment of patients with massive injuries and little chance of survival. The importance of using the concept of triage was reaffirmed by both the British and the Americans in the two operations.

The appropriateness and importance of establishing echelons of medical support was also reaffirmed in both the Falklands Campaign and the Grenada Expedition. The British used every echelon of care effectively and were very careful to coordinate close ties and working relationships between

medical units. The American operational planners, however, initially excluded corps or third level units from the Grenada Expedition. Miscommunication at the operational level incorrectly led the 82nd Airborne Division commander to expect division and corps level medical support to be provided by the U.S. Navy. Although the Navy did eventually provide some medical support equivalent to these levels, the initial lack of planning for the third level of medical care left a serious gap in the medical support system.

The elements of combat medicine, as stated in Chapter Three, provide a common framework used by both the British and the Americans in providing medical support. This operational concept is grounded in two facts: first, that first and second echelon medical care is usually given in a tactical environment and must, therefore, be relatively austere; and second, that most soldiers who will die from wounds will do so within four to six hours if they are not treated.

The British knew these facts full well and planned their medical support with them in mind. Although the terrain in the Falklands and the resulting emphasis on night fighting caused a number of violations of this concept, the British, however, were basically successful in this area except for their failure to use tourniquets. Similarly, the Americans knew those facts and, even though they had hardly any time to plan, were also basically successful in providing advanced trauma life support within the acceptable timeframe.

Patient evacuation, however, was a weakness of both operations. It was a weakness in only one aspect for the British and that was with respect to helicopter evacuation. Once again, the lack of dedicated medical helicopters left casevac susceptible to the shifting priorities for helicopters in British combat operations. So, patient evacuation was deficient in this aspect of the Falklands medical support.

Patient evacuation was also a weakness for the Americans in Grenada and in several aspects even though the American Army advocates medical control of aerial evacuation and stresses aeromedical evacuation to the maximum extent feasible. First, as mentioned earlier, medical helicopters were not available until the combat operations in Grenada were half over. This resulted in non-medical helicopters being used like the British used them in the Falklands. Second, along with the rest of the disruption in the deployment of medical units, the number of ground ambulances deployed was less than needed to support unit and division level medical evacuation needs. This was the case even though the terrain in Grenada did allow use of ground ambulances. Third, unlike the British in the Falklands who had ambulance ships waiting for patients and an established plan for patient airlift back to Britain, the Americans failed to activate an available Air Force aeromedical evacuation squadron until combat operations were nearly over. Consequently, cargo aircraft had to be commandeered initially

and patients were flown not necessarily to the best place for their treatment but to wherever the planes were going.

Finally, even though the operational concept of command and control has essentially already been addressed in the discussion of the medical support principle of control, it is important to note the significance of command and control specifically in the medical support of a rapid deployment force operation. The extremely limited amount of time available to medical officers in preparing for medical support of a rapidly deployed land force argues for even greater than normal reliance by tactical planners on medical command and control of medical resources. Operational planners and tactical decisionmakers have to realize that medical officers are experts in the medical field and should be allowed to plan medical support and execute medical support operations without interference.

The British medical officers did have to overcome some problems with operational planners and tactical decisionmakers in the Falklands Campaign. Yet, on the whole, command and control of medical resources was left to medical officers. The only aspect of control not available in the Falklands - control of casevac helicopters - has already been discussed and is an aspect of a British medical support operational concept. Except for that, medical command and control worked well within the British Army and between the Army and the other British military services.

As previously stated, the American Army medical

commanders found themselves excluded from the planning process. They were even denied normal participation in the deployment and execution phases because of the confusion about available medical support and the perceived enemy threat on the island. Although it can be argued that this situation developed from the concern for operational security for the Grenada Expedition, this researcher believes that such operational considerations actually argue for greater, not less, participation for medical commanders because medical command and control is even more important than usual in a short-fused, joint operation like Grenada.

### Lessons Learned

Key lessons from the medical support in the Falklands Campaign and the Grenada Expedition have been discussed in the two previous chapters. What will be discussed here are the lessons from those two medical support operations which are particularly relevant for the planning, execution, command and control of future medical support operations of rapidly deployed land forces.

The importance of planning is one of the most vital lessons to be learned from these examples of medical support of combat operations of rapidly deployed land forces. The

British example in the Falklands clearly illustrates effective planning in two dimensions. First, it illustrates the involvement of medical planners by operational planners in the operational planning process. Second, it points out the effectiveness of the planning by medical commanders both within their own service and between services. The British medical officers, like their counterparts in other military branches, efficiently used the limited time available to them and developed a superb plan for the Falklands Campaign medical support. Considering the relatively austere condition of the British Army prior to the Falklands, British planners at every level insured the best operation possible would be conducted.

The importance of planning is also illustrated by the Grenada Expedition. The operational problems there point out that operations planned in such haste must be simple or else run the severe risk of being so complicated they jeopardize success. Certainly the Americans had much less time to plan the Grenada Expedition than the British had to plan their campaign in the Falklands. But even in view of the severe time restrictions, the Americans should have done a better job of involving CSS planners and in devising joint, unified plans for tactical operations and medical support.

Another lesson important for planning future medical support of rapidly deployed land forces concerns operational security. Operational security, though vital to quick strike operations, cannot be allowed to prevent effective, necessary

planning. The operational security for the Grenada Expedition was essentially discarded when the Cubans were warned two hours in advance of the Ranger's landing. Yet, that security, which had kept units with a real need to know out of the picture for days, ultimately served little purpose except to jeopardize the operation it was designed to protect. This lesson is especially worthy to note because the problem with operational security and medical support is a recurring one for rapidly deployed land force operations. For example, the same situation as in the Grenada Expedition occurred in the Lebanon Operation of 1958.<sup>1</sup>

The lack of planning for Grenada medical support and the limited time for planning British medical support in the Falklands also serve to point out the need for standard medical support packages. These packages, when designed with a high degree of intrinsic flexibility, can greatly decrease the time needed to plan medical support of rapid combat deployments. The effectiveness of the 307th Medical Battalion's deployment packages proves this point.

Finally, with respect to planning, the two medical support operations examined in this thesis depict the limits of the tailored medical support spectrum. The Falklands medical support represents a medical support package expressly designed for a specific land force campaign. The Grenada Expedition medical support, on the other hand, represents a medical support package that was essentially an off-the-shelf package designed for general contingency

missions without specific tailoring for the combat operations in Grenada.

There are a number of important medical lessons from Grenada and the Falklands with regard to the execution of medical support of rapid deployment forces. The Falklands Campaign illustrated the significance of physical fitness, self aid, buddy aid, tourniquets, far forward resuscitation, simple surgical procedures, and simple clinical policies. The Falklands also pointed out the need for greater emphasis on fluid intake by soldiers and broader training for military physicians. Grenada also revealed these last two lessons.

Grenada, in a negative sense, revealed the need for joint medical understanding and cooperation in the execution of medical support. It revealed that initiative of medical soldiers can accomplish much, but that joint execution of medical support must not be lacking if soldiers' lives are to be saved.

Grenada also demonstrated the need for a comprehensive medical system to either be in place in the combat zone or readily accessible from it. Proper execution of medical support, especially in a rapid deployment force operation when time is so critical, requires completeness in the medical support system. Without that completeness, patients may die because the time for effective medical intervention has been lost.

Lastly, the lessons about command and control stand out. Even though a few internal problems were revealed in the

medical units of both the British and the Americans in the operations studied, the significant problems encountered would have been eliminated had more medical command and control been allowed. Where medical commanders were allowed to control their support operations as they should, as in the Falklands, medical support was highly effective and efficient. Where medical commanders were circumvented or excluded, as in Grenada, medical support was placed in great jeopardy. Further, the significant role medical commanders played in making timely medical support decisions in the Falklands Campaign and in the Grenada Expedition demonstrates the absolute necessity for active medical leadership, clear medical command, and direct medical control.

These lessons provide answers to the questions about how to effectively meet the medical needs of rapidly deployed land forces. Due to the increasing likelihood of more rapid deployment force operations in the future, these answers are vital. They will enable operational decisionmakers and medical commanders to plan and execute rapid medical support operations with precision so that another vital element for the success of the land force can be assured.

# ENDNOTES

# CHAPTER VI

Evaluation of Medical Service Support for the Lebanon
Operation, Office of the Assistant Secretary of Defense
(Health and Medical), Washington, D.C., 18 February 1960:
3.

### CHAPTER VII

## CONCLUSIONS AND RECOMMENDATIONS

#### Conclusions

The preceding comparative analysis of medical support in the Falklands Campaign and the Grenada Expedition serves to illustrate how medical support of rapidly deployed land forces has been conducted. It also reveals key points for medical commanders to consider in conducting medical support of future rapidly deployed land forces. The medical support operations in the Falklands and Grenada are not the only examples available of medical support of rapidly deployed land forces but, in the researcher's opinion they are the best and the most recent.

The Falklands and Crenada medical support operations have been compared using the principles of conformity, proximity, flexibility, mobility, continuity, and control. These principles, in addition to significant and commonly shared operational concepts, have enabled an examination of the medical support of the two operations.

The commonly shared principles and operational concepts

have served as excellent tools for comparing the two medical support operations studied. They are also excellent tools for planning and executing future medical support operations of rapidly deployed land forces. The lessons derived from the comparison also provide answers to vital questions about how to medically support future combat operations.

#### Recommendations

The six principles which form the basis of both American and British medical support doctrine have been useful in examining the medical support of the combat operations in two rapid deployment force operations - the Falklands Campaign and the Grenada Expedition. The principles' real value, however, is in their use to plan and execute medical support of future rapid deployment force operations.

Medical support planned using the principles of conformity, proximity, flexibility, mobility, continuity, and control will provide the foundation for subsequent successful execution. Execution following the principles will enable medical support of combat operations to be completed effectively and efficiently.

Medical support of rapidly deployed land forces is a key issue for the AMEDD because the likelihood of combat operations by such forces is increasing significantly. The United States, in its role as one of the two world

superpowers, may increasingly be called upon for rapid military intervention to secure areas vital to this nation and its allies. If the AMEDD is to meet its obligation of providing timely medical support to the soldiers conducting these quick operations, medical commanders must enable their organizations to react rapidly and be proactive in preparing their medical support capabilities.

Tactical and operational commanders and their staffs must also recognize the role the AMEDD has to play in achieving the success of their rapid deployment operations. They can be confident that they can rely on their medics to give them the best support available if they ensure that medical planners and medical commanders are involved throughout the planning and execution process of a rapid deployment force operation.

The medical triumphs, as well as the medical failings, of the medical support in the Falklands and Grenada are evident when these rapid deployment operations are analyzed using the established and time-tested medical support principles. Violating any of the principles opens the door to a potential medical disaster and jeopardizes overall operational success.

In addition to the points already discussed, this analysis has revealed the following key recommendations:

(1) soldiers must be highly physically fit for rapid deployment operations and must be encouraged to increase their fluid intake during the operation;

(2) soldiers must be trained in both self aid and buddy aid to augment unit medics especially in the event of a mass casualty situation;

(3) medical soldiers must provide resuscitation of casualties, especially in the control of hemorrhage, as far forward and as quickly as possible;

(4) medical soldiers, especially physicians, must be familiar with the special requirements of battlefield medicine and know their medical equipment and organizational structure;

(5) medical equipment packages must be designed to be lightweight and meet a wide range of medical support contingencies through intrinsic flexibility and functional integrity;

(6) medical support relationships must be clearly defined and understood by supported units and between medical units of the same as well as other services;

(7) the patient evacuation system must be clearly defined and established concurrently with the deployment of combat forces to expedite patient movement within a service's medical echelons and between services in joint operations;

(8) patient evacuation must be by air whenever feasible and in dedicated medical aircraft to ensure continuity of care while the patient is enroute to the next echelon of care;

(9) the accessibility of medical care outside of a combat zone must be maintained using a thoroughly coordinated

patient evacuation system or the full spectrum of medical support capabilities must be maintained in the combat zone;

(10) medical units, like all military units, should be deployed with their habitually associated support units whenever possible to insure the fastest medical support of rapid deployment forces;

(11) simplicity must be stressed throughout the battlefield medical support spectrum from simple first aid techniques (like tourniquets) to simple surgical procedures and clinical policies; and

(12) medical commanders must expect unforeseen problems to arise so they should stay flexible and document steps taken to overcome problems so information will be available for future training and planning.<sup>1</sup>

### Recommendations for Further Study

Now that a comparison of the medical support of two recent rapidly deployed land force operations by two different countries has been made, further studies of other medical support operations of rapid deployment forces should be made to determine if different lessons may be learned. Also, studies should be conducted to determine if the principles used by the American and British Armies are shared by other armies and whether any other principles have
been demonstrated to be of significant value in the planning and execution of rapid deployment force medical support. Similarly, medical support operational concepts of other armies should be examined and analyzed in the light of comparisons to actual rapid deployment force operations.

Several other questions surfaced during the research for this study which merit additional study. First, the two operations studied used traditional medical battalion-like organizations to command and control second echelon medical support. Under the U.S. Army's current structure, most divisions have forward support battalions (each with one medical company) and no medical battalion like the 82nd Airborne Division. This study has shown the key role played by the 307th Medical Battalion commander in insuring medical support was deployed on time to prevent a medical disaster. A study should be conducted to determine if forward support battalion commanders or DISCOM level medical staff officers can be expected to intervene at such a critical time and have their medical concerns accepted by tactical commanders.

A second question concerns the fourth category of triage. That category - expectant, which is used for patients with massive injuries and little chance of survival even if all medical resources are concentrated on them was never used in the Falklands or Grenada. Fortunately, neither of those operations resulted in casualty levels which exceeded available medical resources. Nevertheless, the question which must be asked is would loss of life in a rapid

deployment force operation be accepted simply because medical capabilities and resources were insufficient? Certainly in a major campaign large numbers of casualties can be expected and therefore, occasionally, medical officers may have to declare certain patients in that fourth priority. But in a small scale action such as most rapid deployment operations have been, can a medical department afford not to provide sufficient resources to preclude the need for using the expectant category of triage?

Another key question concerns preventive medicine measures. The focus of this thesis has been on the acute care provided to casualties in two combat operations. Ard, although some attention was given to preventive medicine considerations in the narrative, the full importance of preventive medicine measures for rapid deployment forces has not been addressed. There are significant preventive medicine implications in both the Falklands and Grenada even though they were for relatively short periods of time. Even the Grenada Expedition, whose combat operations lasted only 96 hours, had significant preventive medicine problems which received attention by tactical commanders much later than appropriate.

Fourthly, the role of the Navy's new hospital ships should be analyzed to determine their availability and capability to support rapidly deployed land forces. This new medical support resource should be studied in the light of American historical information (use of hospital ships in

World War II) and in comparison to the British experience with their hospital ship in the Falklands. The medical support implications of the U.S.N.S. *Mercy* and the U.S.N.S. *Comfort* are important both for rapid deployment forces and other operational forces.<sup>2</sup>

Finally, a study of a number of medical support operations of rapidly deployed land forces of several different nations would be useful in constructing a matrix. This matrix could be used to compare different military medical departments and their performance with respect to the six medical support principles used in this study. This would be an effective means to compare the actual performance of medical support operations of other nations, especially our allies. Such a study would be useful in assessing the future ability of our allies to medically support American soldiers in campaigns requiring coalition warfare.

# ENDNOTES

# CHAPTER VII

Lieutenant Colonel Kenneth C. Sever, "782nd Maintenance
Battalion in Grenada," <u>Ordnance</u> 2 (Winter 1984): 6.

"New Hospital Ship Will Increase Readiness," <u>HSC Mercury</u>
(February 1987): 1.

SELECTED

٠

•

BIBLIOGRAPHY

#### SELECTED BIBLIOGRAPHY

### <u>Periodicals</u>

#### Falklands

- Adams, Valerie. "Logistic Support for the Falklands Campaign." <u>Journal for Defense Studies</u> 129 (September 1984): 43.
- Bailey, Jonathan, Major, Royal Artillery. "Training for War: The Falklands 1982." 63 <u>Military Review</u> (September 1983): 58.
- Cordesman, Anthony H. "The Falklands Campaign: The Lessons for British Defense Planning." <u>Armed Forces Journal</u> <u>International</u> 120 (February 1983): 22.
- "Despatch by Admiral Sir John Fieldhouse, GCB GBE, Commander of the Task Force Operations in the South Atlantic: April to June 1982 [sic]." Supplement to <u>The London</u> <u>Gazette</u>. (Monday, 14 December 1982): 16109.
- Dopson, L. "Falklands Special: Saving Lives in the South Atlantic." <u>Nursing Times</u> 78 (June 23-29, 1982): 1044.
- "Falklands Postscript." <u>U.S. Naval Institute Proceedings</u> 109 (June 1983): 99.
- Harrison, Sir John, Vice-Admiral. "Naval Medicine in the Falklands Conflict, April - July 1982, Overall Policy and Operations." Transcripts of the Medical Society of London 84 (November 28, 1983): 75.
- Jackson, D.S., Major. "Sepsis in Soft Tissue Limb Wounds in Soldiers Injured During the Falklands Campaign 1982." Journal of the Royal Army Medical Corps 130 (June 1984): 97.
- Jackson, D.S.; Batty, C.G.; Ryan J.M.; and McGregor, W.S. "The Falklands War: Army Field Surgical Experience." <u>Annals of Royal College of Surgeons</u> 65 (September 1983): 281.
- Jackson, D.S.; Jowitt, M.D.; and Knight, R.J. "First and Second Line Treatment in the Falklands Campaign: A Retrospective View." Journal of the Royal Army Medical Corps 130 (June 1984): 79.
- Jowitt, M.D., Major. "Anaesthesia Ashore in the Falklands." <u>Annals of the Royal College of Surgeons</u> 66 (May 1984): 197.

- Jowitt, M.D., and Knight, R.J. "Anaesthesia During the Falklands Campaign -- The Land Battles." <u>Anaesthesia</u> 38 (August 1983): 776.
- Kenney, David J., Captain. "The Fascinating Falklands Campaign." <u>U.S. Naval Institute Proceedings</u> 109 (June 1983): 100.
- "Lesson of Falklands: Prepare for Surprises." <u>U.S. Medicine</u> (February 1, 1983): 3.
- Lewis, Brenda R. "The Falklands War A Recap." <u>Army</u> 32 (September 1982): 22.
- Marsh, A.R. Commander. "A Short But Distant War The Falklands Campaign." Journal of the Royal Society of <u>Medicine</u> 76 (November 1983): 972.
- Milton-Thompson, G.J. "The Falklands Campaign A Medical Overview." <u>Journal of the Royal Naval Medical</u> <u>Service</u> 69 (Spring 1983): 6.

ロートからのたる

- Murguizur, Dr. Juan Carlos. "The South Plantic Conflict: An Argentinian Point of View." <u>International Defense</u> <u>Review</u> 16 (November 2, 1983): 135.
- Nott, John. "The Falklands Campaign." <u>U.S. Naval Institute</u> <u>Proceedings</u> 109 (May 1983): 118.
- Parry, E.S. Lieutenant Colonel. "Provision of Blood during the Falklands Campaign." <u>Journal of the Royal</u> <u>Society of Medicine</u> 77 (April 1984): 343.
- Price, H.H., Captain. "The Falklands: Rate of British Psychiatric Combat Casualties Compared to Recent American Wars." Journal of the Royal Army Medical Corps 130 (June 1984): 109.
- Raggett, Robert J. "Building Better C<sup>3</sup> I Systems." <u>Signal</u> 38 (November 1983): 73.
- Ratcliffe, G.E.; Cetti, N.E.; and Bleany, A.A. "Medicine in the Falklands." Journal of the Royal Army Medical Corps 130 (February 1984): 16.
- Reed, John. "Commando Logistic Regiment Royal Marines." <u>Armed Forces</u>. (June 1983): 212.
- Richards, Tessa. "Medical Lessons from the Falklands." British Medical Journal 286 (March 5, 1983): 799.
- Ryan, J.M., Major. "The Falklands War Triage." <u>Annals of</u> the Royal College of Surgeons 66 (May 1984): 195.

- "Saving Lives in the South Atlantic." <u>Nursing Times</u> 78 (June 23, 1982): 1044.
- Sherman, Elizabeth J. "In Bitter Little Falklands War, Enemy Medics Cooperated." <u>Army Times</u> 20 (December 29, 1986): 11.
- Summers, Harry G. "Yomping to Port Stanley." <u>Military</u> <u>Review</u> 64 (March 1984): 2.

•

- Trotter, Neville. "The Falklands Campaign -- Command and Logistics." <u>Armed Forces Journal International</u> 120 (June 1983): 32.
- Whittaker, R.N. "Combat Logistics: The South Atlantic." Armed Forces Journal of Logistics 8 (Fall 1984): 2.

#### <u>Grenada</u>

an bar fim en men eine aufent men meller eine fichten in einen einen verein vereinen einen einen

- Bennett, Ralph Kinney. "Grenada: Anatomy of a 'Go' Decision." <u>Reader's Digest</u> (February 1984): 72.
- Byron, Michael J., Lieutenant Colonel. "Fury from the Sea: Mrg Los in Grenada." <u>U.S. Naval Institute</u> <u>Prophy</u> ings 110 (May 1984): 119.
- Cragg, Dan. "The U.S.Army in Grenada." <u>Army</u> 33 (December 1983): 29.
- Dubik, James M., Major; and Fullerton, Terrence D., Major. "Soldier Overloading in Grenada." <u>Military Review</u> 67 (January 1987): 38.
- Duffy, Michael. "Grenada: Rampant Confusion." <u>Military</u> Logistic Forum 2 (July-August 1985): 20.
- Emerson, Steven. "What Went Wrong on Grenada?" <u>U.S. News</u> and <u>World Report</u> 101 (November 3, 1986): 42.
- Fenstermacher, Willard B. "Capstone Relationship Put to Test in Grenada." <u>Officer</u> 60 (February 1984): 21.
- Gabriel, Richard. "Scenes from an Invasion." <u>The Washington</u> <u>Monthly</u> (February 1986): 34.
- "JOS Replies to Criticism of Grenada Operations." <u>Army</u> 34 (August 1984): 28.
- Magnuson, Ed. "D-Day in Grenada." <u>Time</u> 122 (November 7, 1983): 22.
- Magnuson, Ed. "Now to Make It Work." <u>Time</u> 122 (November 14, 1983): 18.
- Manning, Robert A.; Emerson, Steven; and Kaylor, Robert. "Special Forces: Can They Do the Job?" <u>U.S. News and</u> <u>World Report</u> 101 (November 3, 1986): 36.
- Maze, Rick. "Intelligence 'Limited' Before Grenada Action." <u>Air Force Times</u> 44 (November 21, 1983): 4.
- "Operation Island Breeze Was Not a Breeze." <u>Army Logistician</u> 16 (November-December 1984): 20.
- Perkins, Andrew M., Jr., Lieutenant Colonel. "Operation Urgent Fury: An Engineer's View." <u>The Military</u> <u>Engineer</u> 76 (March-April): 86.
- Radin, Robert M., Captain; and Bell, Raymond A., Chief Warrant Officer. "Combat Service Support of Urgent Fury." <u>Army Logistician</u> 16 (November-December 1984):

16.

- Schafer, Charles H. "Logistics Support of the Caribbean Peacekeeping Force (CPF)." <u>Army Logistician</u> 17 (September-October 1985): 22.
- Sever, Kenneth C., Lieutenant Colonel. "782nd Maintenance Battalion in Grenada." <u>Ordnance</u> 2 (Winter 1984): 4.
- Wolcott, Barry W., Colonel. "Combat Medicine." <u>Military</u> <u>Review</u> 67 (January 1987): 31.

### Additional Periodical Sources

. . . .

- Bellamy, Ronald F., Colonel. "The Causes of Death in Conventional Land Warfare: Implications for Combat Casualty Care Research." <u>Military Medicine</u> 149 (February 1984): 55.
- Cheney, Congressman Richard B., and Harvey, Thomas N., Major(P). "Strategic Underpinnings of a Future Force." <u>Military Review</u> 66 (October 1986): 4.
- "Copter Ambulances Improve Survival Rate." <u>The Journal of</u> <u>the Armed Forces</u> 103 (December 11, 1965): 12.
- Dolev, Eran, and Llewellyn, Craig, Colonel. "The Chain of Medical Responsibility in Battlefield Medicine." <u>Military Medicine</u> 150 (September 1985): 471.
- Hardaway, Robert M., III, Brigadier General. "Contributions of Army Medicine to Civilian Medicine." <u>Military</u> <u>Medicine</u> 138 (July 1973): 409.
- Heaton, Leonard D., Lieutenant General. "Medical Support in Vietnam." <u>Army</u> 16 (October 1966): 125.
- Heaton, Leonard D., Lieutenant General. "Medical Support of the Soldier: A Team Effort in Saving Lives." <u>Army</u> 19 (October 1969): 85.
- Jones, Ellis L.; Peters, Albert F., Captain; and Gasior, Robert M., Captain. "Early Management of Battle Casualties in Vietnam." <u>Archives of Surgery</u> 97 (July 1968): 1.
- Motley, James B., Colonel. "Washington's Big Tug of War Over Special Operations Forces." <u>Army</u> 36 (November 1986): 16.
- Neel, Spurgeon, Colonel. "Army Aeromedical Evacuation Procedures in Vietnam." Journal of the American Medical Association 204 (April 22, 1968): 99.
- "New Hospital Ship Will Increase Readiness." <u>HSC Mercury</u> 14 (February 1987): 1.

#### Books

Falklands

- Bishop, Patrick; and Witherow, John. <u>The Winter War: The</u> <u>Falklands</u>. New York: Quarter Books, Inc., 1983.
- Calvert, Peter. <u>The Falklands Crisis: The Rights and the</u> <u>Wrongs</u>. New York: St. Martin's Press, 1982.
- Fox, Robert. Eyewitness Falklands. London: Methuen, 1982.
- Frost, John. <u>2 PARA Falklands: The Battalion at War</u>. London: Buchan and Enright, Publishers, Limited, 1983.
- Hastings, Max; and Jenkins, Simon. <u>The Battle for the</u> <u>Falklaids</u>. New York: W.W. Norton and Company, 1983.
- Hoffmann, Fritz L.; and Hoffmann, Olga Mingo. <u>Sovreignty in</u> <u>Dispute: The Falklands/Malvinas, 1493-1982</u>. Boulder, Colorado: Westview Press, 1984.
- Jolly, Rick. <u>The Red and Green Life Machine: A Diary of the</u> <u>Falklands Field Hospital</u>. London: Century, 1983.
- Spanier, John. <u>James Nations Play</u>. New York: Holt, Rinehart and Winston, 1984.
- <u>The Sunday Times of London Insight Team. War in the</u> <u>Falklands: The Full Story</u>. New York: Harper and Row, Publishers, Inc., 1982.
- Watson, Bruce W.; and Dunn, Peter M., ed. <u>Military Lessons</u> of the <u>Falklands Islands War:</u> <u>Views from the United</u> <u>States</u>. Boulder, Colorado: Westview Press, 1984.

# <u>Grenada</u>

. .....

- Dunn, Peter M.; and Watson, Bruce W. ed. <u>American</u> <u>Intervention in Grenada: The Implications of</u> <u>Operation Urgent Fury</u>. Boulder, Colorado: Westview Press, 1985.
- Gabriel, Richard A. <u>Military Incompetence: Why the American</u> <u>Military Doesn't Win</u>. New York: Farrar, Straus and Giroux, 1985.
- O'Shaughnessy, Hugh. <u>Grenada: Revolution, Invasion and</u> <u>Aftermath</u>. Great Britain: Sphere Books, Ltd., 1984.
- Payne, Anthony; Sutton, Paul; and Thorndike, Tony. <u>Grenada:</u> <u>Revolution and Invasion</u>. New York: St. Martin's Press, 1984.
- Schoenhals, Kai P.; and Melanson, Richard A. <u>Revolution and</u> <u>Intervention in Grenada: The New Jewel Movement, the</u> <u>United States, and the Caribbean</u>. Boulder, Colorado: Westview Press, 1985.

# Governmental Publications

- "Administration in War." Royal Army Manual on Administration and Logistics. Army Code Number 71342. Published by the Ministry of Defence, 1984.
- Bishop, Charles R., Major, and O'Brien, E. Kathleen. <u>FORSCOM</u> <u>ARLANT Participation in Operation Urgent Fury --</u> <u>Grenada, 1983</u>. Military History Office, Office of the Chief of Staff, U.S. Army Forces Command, Fort McPherson, Georgia, 1 April 1985.
- Evaluation of Medical Service Support for the Lebanon Operation. Office of the Assistant Secretary of Defense, Washington, D.C. 18 February 1960.
- "The Falklands Campaign: The Lessons." Presentation by Secretary of State for Defence to Parliament. London: Her Majesty's Stationery Office, December 1982.
- <u>The Falklands Campaign: The Lessons (United Kingdom White</u> <u>Paper)</u>. London, England, December 1982.
- "Falkland Islands Campaign Medical Aspects." Briefing Script prepared by the Royal Army Medical Corps presented to Dr. Robert H. Mosebar, Medical Officer, Directorate of Combat Development, Academy of Health Sciences, Fort Sam Houston, Texas.
- <u>Grenada: A Preliminary Report</u>. Department of State and the Department of Defense, Washington, D.C. December 16, 1983.
- <u>Grenada Documents: An Overview and Selection</u>. Department of State and the Department of Defense, Washington, D.C., September 1984.
- Grenada Work Group. <u>Operation Urgent Fury Assessment</u>. Combined Arms Center, U.S. Army Training and Doctrine Command. Fort Leavenworth, Kansas, 9 July 1985.
- Harmon, John W., Lieutenant Colonel, and Lewellyn, Craig, Colonel. Report on the Symposium on the Falkland Islands Campaign - Medical Lessons at the Royal College of Surgeons, London, England, 17-18 February 1983.
- Heiser, Joseph M., Jr., Lieutenant General. <u>Vietnam Studies:</u> <u>Logistics Support</u>. Washington, D.C.: Department of the Army, 1974.
- Lessons Learned Grenada: U.S. Army Lessons Learned From 1983 Operation Urgent Fury. Washington, D.C.: Department

of the Army, 16 May 1985.

- "Logistics 1". Precis from the United Kingdom Army Staff College in files of British Liaison Officer, U.S. Army Command and General Staff College.
- "Logistics 2". Precis from the United Kingdom Army Staff College in files of British Liaison Officer, U.S. Army Command and General Staff College.
- "Medical Interoperability Handbook." Royal Army Manual Code Number 71376. Published by the Ministry of Defence, February 1986.
- "Medical Support in the Field." Precis from the Royal Army Medical Corps Training Centre in files of British Liaison Officer, U.S. Army Surgeon General's Office.
- Mosebar, Robert H., M.D. "Lessons Learned in Lebanon and the Falklands." Briefing script provided by the author on 6 January 1987.
- Mosebar, Robert H., M.D. "Field Hospital in the Falkland Islands." Memorandum prepared by the Medical Officer, Directorate of Combat Development, Academy of Health Sciences, Fort Sam Houston, Texas, 18 January 1984.
- Motley, Langhorne A. "The Decision to Assist Grenada." U.S. Department of State Current Policy No. 541, Washington, D.C., 24 January 1984.
- Neel, Spurgeon, Major General. <u>Vietnam Studies: Medical</u> <u>Support of the U.S. Army in Vietnam 1965-1970</u>. Washington, D.C.: Department of the Army, 1973.
- Pirnie, Bruce R., Major, U.S. Army. <u>Operation Urgent Fury:</u> <u>The United States Army in Joint Operations</u>. (CMH Publication 93-4). Analysis Branch, U.S. Army Center for Military History, Washington, D.C., 1986.
- Privratsky, Kenneth L., Major. <u>British Combat Service</u> <u>Support on East Falkland: Considerations for</u> <u>Sustaining Tactical Operations in Remote Areas</u>. School of Advanced Military Studies, U.S. Army Command and General Staff College. Fort Leavenworth, Kansas, 25 November 1985.
- Raffle, John R., Jr., Lieutenant Colonel. <u>Combat Service</u> <u>Support: Introduction to Combat Service Support</u> <u>S480/6</u>. U.S. Army Command and General Staff College, Fort Leavenworth, Kansas, 13 May 1985.
- <u>Strategic Studies: National Security Policy Considerations</u>. U.S. Army Command and General Staff College, Fort

Leavenworth, Kansas, 1986.

- U.S. Department of the Army Field Circular 8-15-1. "Health Service Support Operations-Light Infantry Division." 1 August 1985.
- U.S. Department of the Army Field Circular 8-15-2. "Health Service Support Operations-Airborne and Air Assault Division." 1 January 1986.
- U.S. Department of the Army Field Circular 8-45. "Medical Evacuation in the Combat Zone." 10 October 1986.
- U.S. Department of the Army Field Manual 8-8. "Medical Support in Joint Operations." 1 June 1972.
- U.S. Department of the Army Field Manual 8-10. "Health Service Support in a Theater of Operations." 2 October 1978.
- U.S. Department of the Army Field Manual 8-20(Test). "Health Service Support in a Combat Zone." 31 May 1983.
- U.S. Department of the Army Field Manual 8-35. "Evacuation of the Sick and Wounded." 22 December 1983.
- U.S. Department of the Army Field Manual 8-55. "Planning for Health Service Support." 15 February 1985.
- U.S. Department of the Army Field Manual 63-2. "Combat Service Support Operations-Division." 21 November 1983.
- U.S. Department of the Army Field Manual 63-3J. "Combat Service Support Operations-Corps." 12 August 1985.
- U.S. Department of the Army Field Manual 63-4. "Combat Service Support Operations-Theater Army Area Command." 24 September 1984.
- U.S. Department of the Army Field Manual 100-5. "Operations." 5 May 1986.
- U.S. Department of the Army Field Manual 100-10. "Combat Service Support." 1 March 1983.
- U.S. Department of the Army Regulation 351-3. "Professional Training of AMEDD Personnel." 15 December 1985.
- U.S. Department of the Army Training and Doctrine Command Pamphlet 525-50 "U.S. Army Operational Concept for Health Service Support AirLand Battle." 11 April 1986.

U.S. Department of the Army White Paper. "Health Service Support AirLand Battle." 11 December 1986.

Wade, Gary H., Lieutenant Colonel. <u>Rapid Deployment</u> <u>Logistics: Lebanon 1958</u>. Combat Studies Institute, U.S. Army Command and Staff College. Fort Leavenworth, Kansas, October 1984.

#### Additional Sources and Interviews

i

111111

i

- Ashley, Vincent E., Captain. Academy of Health Sciences, U.S. Army Health Services Command, Fort Sam Houston, Texas. Letter interview. 8 December 1986. (Captain Ashley commanded Company C, 307th Medical Battalion, supporting the 2d Brigade, 82nd Airborne Division. during the Grenada Expedition.)
- Bradley, Stephen S., Captain. U.S. Army Medical Department Activity, West Point, New York. Letter interview, 31 December 1986. (Captain Bradley commanded Company A, 307th Medical Battalion, supporting the 3d Brigade, 82nd Airborne Division, during the Grenada Expedition.)
- Creamer, Ian, Colonel. British Liaison Officer, Royal Army Medical Corps, to the U.S. Army Surgeon General, Washington, D.C. Telephone interview, 2 March 1987.
- Heard, R.J.B., Colonel. British Liaison Officer, U.S. Army Command and General Staff College, Fort Leavenworth, Kansas. Interview, 24 March 1987.
- Mosebar, Robert H., M.D. Medical Officer, Directorate of Combat Development, Academy of Health Sciences, U.S. Army Health Services Command, Fort Sam Houston, Texas. Telephone interviews, 9 February 1987 and 10 March 1987.
- Nolan, David L., Major. Health Care Operations, Office of the Surgeon General, United States Army, Washington, D.C. Telephone interview, 22 December 1986. (Major Nolan was the 307th Medical Battalion's operations officer during the Grenada Expedition.)
- Phelps, Douglas S., Captain. Combined Arms Services Staff School, Fort Leavenworth, Kansas. Interview, 4 March 1987. (Captain Phelps was the executive officer of Company C, 307th Medical Battalion - the first medical company deployed on the Grenada Expedition.)
- Phelps, Douglas S., Captain. "Synopsis of Division Level Medical Support in Grenada." Undated after action summary provided by the author during the 4 March 1987 interview.
- Swenie, Kevin J., Major. Health Care Operations, Office of the Surgeon General, United States Army, Washington, D.C. Telephone interview, 5 March 1987. (Major Swenie was the operations officer for the 57th Medical Detachment (Air Ambulance) during the Grenada Expedition.)

Wilson, Edward B., Lieutenant Colonel. Health Care Operations, Office of the Surgeon General, United States Army, Washington, D.C. Telephone interview, 23 December 1986. (Lieutenant Colonel Wilson commanded the 307th Medical Battalion, 82nd Airborne Division, during the Grenada Expedition.)

## INITIAL DISTRIBUTION LIST

Combined Arms Research Library U.S. Army Command and General Staff College Ft. Leavenworth, Kansas 66027

Defense Technical Information Center Cameron Station Alexandria, Virginia 22314

LTC Scottie T. Hooker Combined Arms Services Staff School U.S. Army Command and General Staff College Ft. Leavenworth, Kansan 66027

LTC Nick L. Johnson Department of Sustainment and Resourcing Operations U.S. Army Command and General Staff College Ft. Leavenworth, Kansas 66027

COL Max G. Manwaring PSC Box 1678 APO Miami, FL 34003

Commandant Academy of Health Sciences ATTN: HSHA-DCD Ft. Sam Houston, Texas 78234-6100

Commander 2d Medical Battalion 2d Infantry Division APO San Francisco 96224

Commander 7th Medical Battalion 7th Infantry Division (Light) Ft. Ord, California 93941-6060

Commander 10th Medical Battalion 10th Mountain Division (Light) Ft. Drum, New York 13602-5000

Commander 25th Medical Battalion 25th Infantry Division (Light) Schofield Barracks, Hawaii 96225-6000 Commander 307th Medical Battalion 82d Airborne Division Ft. Bragg, North Carolina 28307-5100

Commander 326th Medical Battalion 101st Airborne Division (Air Assault) Ft. Campbell, Kentucky 42223-5000

Commander 142nd Medical Battalion 193d Infantry Brigade APO Miami 34004-5000

